

Received 25 December 2022, accepted 10 January 2023, date of publication 17 January 2023, date of current version 25 January 2023. Digital Object Identifier 10.1109/ACCESS.2023.3237735

RESEARCH ARTICLE

Solving the Sustainable Automobile **Production-Distribution Joint Optimization in the Physical Internet-Enabled Hyperconnected Order-to-Delivery System by I-NSGAIII**

YAOTING XUE^{^{[D],2}, SHOUFENG JI¹, GUOSONG ZHU², AND PENGYUN ZHAO¹}

¹Department of Operation and Logistics Management, School of Business Administration, Northeastern University, Shenyang 110167, China ²Department of Supply Chain and Transport Planning, BMW Brilliance Automotive Ltd., Shenyang 100027, China Corresponding author: Shoufeng Ji (sfji@mail.neu.edu.cn)

This work was supported by the National Natural Science Foundation of China, China, under Grant 71971049 and Grant 71572031.

ABSTRACT The Physical Internet (PI)-enabled hyperconnected order-to-delivery system (OTD) provides new solutions for sustainable supply chains from production perspectives. In this system, a PI-enabled hyperconnected manufacturing system is more closely tied with other functions through Internetof-Things (IoT)-enabled machines for communication. In the OTD, the PI-enabled hyperconnected production-distribution system (PI-H) is modelled by multi-objective mixed-integer-nonlinear programming to evaluate sustainability. We develop an improved reference-point based non-dominated sorting genetic algorithm (I-NSGAIII) to solve practical-scale PI-enabled hyperconnected production-distribution scheduling problems, with the problem-specific solution expression and dynamic programming, subproblem-guided crossover and mutation strategies, and adaptive evolution mechanisms. I-NSGAIII's performance advantages and PI-H's sustainable advantages are validated through extensive experiments.

INDEX TERMS Integrated production-distribution scheduling, multi-objective optimisation, physical internet, supply chain management, sustainability.

I. INTRODUCTION

For auto enterprises, implementing sustainable supply chain management (SSCM) within the framework of government regulations on environmental policies has the highest priority [4]. The order-to-delivery system (OTD) serves as a supply chain platform, connecting the market and manufacturer. This platform helps manage the production status and delivery progress of orders in production and transportation systems [2]. Hence, sustainably producing and delivering customised vehicles to customers in shorter delivery times is the primary purpose of OTD. Meanwhile, the Internet-of-Things (IoT) revolution promotes the Physical Internet (PI) paradigm in OTD, where the end-to-end visibility of the PI objects, operations and systems is realised [3]. A Physical

The associate editor coordinating the review of this manuscript and approving it for publication was Shih-Wei Lin¹⁰.

Internet-enabled hyperconnected OTD brings a new application scenario of a digital-driven manufacturing system that is more conducive to achieving sustainability in jointly scheduling production and distribution.

Auto supply chains with highly modular production processes have huge potential in digital supply chain implementation [2]. Ji et al. [4] studied the application of PI and found that a PI-enabled auto supply chain (PI-S) can provide a better service level with less cost compared to a traditional auto supply chain (TS), without considering sustainability. However, their study only investigated the advantages of transparency from the tracking function of PI, owing to existing limitations in the infrastructure and ignoring the hyperconnected distribution brought by PI. PI can digitally connect and share information in real-time among physical machines, assets and components, generating the operation scenario with an IoT-enabled machine-to-machine (IoT-M2M) [5]. Our article formulates a model to evaluate the sustainability performance of the operation scenario in IoT-M2M, including the M2M-based automated modular roll spraying system and the hyperconnected distribution. In detailed, the M2M-based automated modular roll spraying system with multiple nozzles inside the robot arm is adopted to form the digital-driven painting workshop(PT). Colours in the nozzle constitute a colour package that massively changes the production logic and human participation compared with the machine-driven PT. Additionally, the hyperconnected distribution of resource flows between plants provides more flexibility for the auto enterprise, requiring unique production facility configurations and costly large-scale equipment [6]. By implementing the PI-enabled hyperconnected production-distribution system (PI-H), these factors promote the formation of hyperconnected OTD for auto enterprises. Moreover, many scholars have conducted extensive qualitative analysis, arguing that PI-M2M will fully and optimally leverage machines while liberating the workforce and improving resource utilisationwhich are two key sustainability indicators [7]. Hence, the potential positive influence brought by PI-H on sustainability is quite worthy of study. As defined in the Johannesburg Summit, sustainable development should balance not just economy and environmental protection but also social development [8]. These gaps between practical requirements of sustainability management in the auto supply chain and existing research allow us to explore implementable PI-H and further investigate its sustainability advantages from economic, environmental and social dimensions.

The integrated production-distribution scheduling (IPDS) has been extensively studied in pursuing a more economical supply chain. Moreover, the interconnection, dynamics and openness brought by PI make IPDS has the characteristic of hyperconnection, w.r.t, PI-HPDS. In this article, the PI-HPDS is formulated as a multiple-objective model to comprehensively evaluate its sustainability from the economic, environmental, and social aspects. Moreover, an I-NSGAIII is proposed to solve the practical-scale problem based on the characteristics of production and distribution in PI-HPDS. The PI-H, PI-S, and TS are investigated under different make-to-order market (MTO) scenarios, combining different customer sizes and market structures. Notably, PI-H has superior performance on sustainability in all instances compared to PI-S and TS. Furthermore, PI-H can intensify these advantages over PI-S as market structures' complexity increases. Finally, the practical value of adopting PI-H for auto-enterprises is discussed. The contributions of this article are summarised as follows:

- The PI-enabled hyperconnected production-distribution system is modelled as MOMINLP to assess its sustainability performance. The M2M-based modular roll spraying system and hyperconnected distribution of resource flow provide a new production scenario for the PI-HPDS.
- We develop the I-NSGAIII to solve practical-scale problems more effectively. In detail, problem-specific

solution expression and dynamic programming, subproblem-guided crossover and mutation operators are proposed. In addition, its performance is verified via statistical comparisons.

• Under the different market structures and demand scales which are the practical situation in MTO, the sustainability performance of PI-H is quantitatively investigated, and some managerial implications are established.

The remaining contents are structured as follows. Section II summarises relevant literature. Section III provides a detailed description of the model. Section IV describes the I-NSGAIII algorithm. Section V performs extensive computational experiments to discuss the efficiency of I-NSGAIII and the performance of the PI-enabled hyperconnected supply chain. Finally, Section VI presents our conclusions and some directions for future work.

II. LITERATURE REVIEW

Current research on jointly scheduling production and distribution prioritises the sustainable performance of the economy, environment and society instead of previous research that has focused on the economy alone. Chandra and Fisher [9] were the first to discuss the huge economic advantage brought by IPDS. Subsequently, IPDS only targeting economic optimisation has been extensively studied under different production configurations. Some researchers assumed that production is performed by a single machine [10]. IPDS with a parallel machine configuration, wherein a job is assigned to multiple machines, has also been studied in [11]. Moreover, Yagmur and Kesen [12] and Homayouni, et al. [13] studied IPDS with a more complex production configuration (e.g. single-stage flow-shop and job-shop). Ji et al. [4] further investigated the cost optimisation of IPDS with a multi-stage manufacturing system (MMS). Through a limited buffer system, jobs are then transferred sequentially in two workshops. Jobs cannot be produced in subsystem i + 1 until subsystem i completes the production and transfer of related jobs. The above studies only focus on the economy. Subsequently, Aghajani and Al-E-Hashem [14], Al-E-Hashem et al. [15], Solina and Mirabelli [16] study how IPDS can consider both environmental and economic goals. Particularly, Al-E-Hashem et al. [15] only focused on carbon emissions in transportation, while Aghajani and Al-E-Hashem [14] and Solina and Mirabelli [16] further studied energy consumption on a single production line. Peng et al. [22] evaluated the sustainability performance of PI-enabled production-inventory-distribution system from economic, environmental and social dimensions. They found that PI will bring better sustainability performance for the supply chain. However, they did not extensively study the social impact of production (e.g. the worker's rights from worker safety and fair production), which is receiving increasing attention. The aforementioned research gaps drive us to explore the sustainability performance of PI-enabled IPDS in more practical production scenarios.

Montreuil [17] developed the concept of PI, a physical metaphor for the digital internet. PI promotes the interconnection and openness of independent supply networks to obtain significant advantages [17]. Moreover, Crainic and Montreuil [18] emphasised hyperconnectivity in logistics and supply chains. Ben Mohamed et al. [19] optimised interconnected urban logistics with economic and ecological attributes. Moreover, Chargui et al. [20] and Hu et al. [21] studied the interconnected distribution system. Peng et al. [22] and Ji et al. [23] successively evaluated PI-enabled production-inventory-distribution system based on cost and sustainability performance. Regardless, the above research simplified the production process as the order allocation problem. Ji et al. [4] further studied IPDS with PIenabled MMS production configuration, which considers job scheduling in sequentially connected two workshops. However, in this article, the two plants independently perform production, and the significance of production for sustainable management is not explored. Furthermore, Marcotte, Montreuil and Coelho [6] firstly introduced the hyperconnected mobile production system. A PI-enabled hyperconnected production system not only relocates processing modules based on demand but also dynamically transfers resources between open plants. Notably, digitally driven manufacturing brought by IoT-M2M in Industry 4.0 considerably changes production scenes in existing MMS and open plants, including the M2M-based modular roll spraying system and the hyperconnected semi-finished delivering system. The research gaps between current research and practical requirements for auto enterprises motivated us to study PI-HPDS based on the PIenabled hyperconnected production-distribution system and evaluate its sustainability.

In recent years, a lot of approaches have been developed to address IPDS, such as the exact method [24] and heuristic algorithms (e.g. adaptive genetic algorithm [10], memetic algorithm [12] and NSGAIII [25]). Exact methods are not undeniably feasible when the problem is scaled up to a certain level. Moreover, most practical problems for enterprises involve multi-objective optimisation. Hence, multiobjective evolutionary algorithms (MOEA) have become suitable solution methods. MOEA based on the Pareto concept has been considered more effective compared to the other methods, including the weighted sum of objective functions and approaches based on population [26]. For instance, strength Pareto evolutionary algorithms (SPEA and SPEA2) have been exposed [27], and non-dominated sorting genetic algorithms (NSGA and NSGA-II) are proposed [28]. Deb and Jain [29] propose the reference-point-based NSGAIII following the NSGA-II framework, which is much better than a classical generating method in many-objective problems. However, NSGAIII does not perform well in largescale optimisation problems. Wang [30] combines several information feedback models with NSGAIII to improve its ability to solve large-scale optimisation problems. However, the above works disregard the mostly nonlinear nature of real problems. In nonlinear problems, the minimal evolutionary

fluctuations in the decision space will cause great distribution changes in the target space. The information entropy theory is suited for solving the PI-HPDS. Table 1 summarises the research gaps in the related literature.

III. MODEL FORMULATION

A. PROBLEM DEFINITION

The proposed PI-HPDS can realise hyperconnected production between two PI-enabled plants and then deliver finishedvehicles to retailers through a multi-level hyperconnected distribution network (Figure 1). The green section describes the hyperconnected production system. Every PI-enabled plant has multiple workshops. Semi-finished products can be delivered between two plants to complete the remaining steps of the production process. Hereafter, only finished-vehicles, which are off the assembly line, will be delivered to customers through a two-level PI-enabled hyperconnected distribution system with RFID, WSN and GPS to perform tracking and tracing. After the finished vehicle leaves the plant, it can be transported to multiple PI-hubs or directly to the retailer. The blue section illustrates the first layer in the transportation system, where trains provide service between plants and hubs. The second layer, marked in yellow, covers routing between PI-hubs and retailers. The PI-HPDS can be defined as a directed network G = (V, E). While V is the vertex set, and E is the edge set, represent arcs between nodes in the vertex set V. Vertex set V includes the plant set P, virtual stock point set P', hub set H and retailer set R. The virtual stock point set P' and hub set H constitute $N_1 = P' \cup H$. M is the set of transporters, $M = \{1, 2\}, m \in M \cap \{1\}$ representing the mode of trains; $m \in M \cap \{2\}$ means the mode of trucks, serving the remaining routing $V \setminus (P \cup H)$. The orders $o \in \{1, 2, ..., |O|\}$ have the following properties: series $K(k \in \{1, 2, ..., |K|\})$, colour $E(e \in \{1, 2, ..., |E|\})$, key part $G(g \in \{1, 2, ..., |G|\})$ and delivery deadline DE_0 . The time horizon is $T(t \in \{1, 2, \dots, |T|\})$, each t in T is the uniformly sized discrete period, and the length of each unit time period can be user-defined when doing time discretisation.

In PI-HPDS, Original equipment manufacturers (OEMs) perform the hyperconnected semi-finished transport for car-shells before the painting shop (PT). Orders are initially pre-allocated to different plants based on the practical constraints of the press workshop only. Hence, the body-inwhite in Plant A will be sent to Plant B for painting and assembly under the whole process of tracking and accurate identification. Meanwhile, the digital painting workshop adopts an M2M-based modular roll spraying system, which has multiple nozzles inside the robot arm. In this study, the total number of spray nozzles $\kappa_n = 3$ and colours in spray nozzles constitute a colour package. If the colour required at the current tact is contained by the colour package, changing the nozzle becomes unnecessary. Notably, spray nozzles can be replaced one or two at a time if they do not work in the current tact. When two spray nozzles are replaced in the same

TABLE 1. Overview of the related literature.

Author	M- SPS	Econo my	Environ mental	Social in production	Social in transportation	PI	Multi- objective	Other features
Chandra and Fisher 2017	-	√	-	-	-	-	-	Multiple products
Low, C. <i>et al.</i> 2017	-	\checkmark	-	-	-	-	-	Single machine
Devapriya <i>et al.</i> 2017	-	\checkmark	-	-	-	-	-	Perishable product
Yagmur and Kesen 2020	-	✓	-	-	-	-	-	Single-stage flow- shop
Homayouni, et al. 2020	-	√	-	-	-	-	-	Single-stage job- shop
Ji et al. 2022	\checkmark	\checkmark	-	-	-	✓	\checkmark	Traditional painting system
Al-E-Hashem <i>et</i> al. 2020	-	✓	~	-	-	-	~	Uncertain demand
Aghajani and Al-E-Hashem 2019	-	\checkmark	\checkmark	-	-	-	\checkmark	Furniture company
Solina and Mirabelli 2021	-	\checkmark	\checkmark	-	-	-	-	Perishable product
Ji <i>et al</i> . 2018	-	\checkmark	-	-	-	-	-	Time window
Peng et al. 2019	-	\checkmark	\checkmark	-	\checkmark	~	~	Ignore social performance in production
Mohamed et al. 2017	-	\checkmark	-	-	-	\checkmark	-	Interconnected city logistics
Chargui <i>et al.</i> 2020	-	✓	-	-	-	~	-	Road-Rail PI- enabled hub PI-enabled
Our study	~	√	√	✓	~	~	V	hyperconnected production distribution system

tact, pollution emission becomes less than that of two nozzles replaced separately in a different tack. Otherwise, at least one of the spray nozzles will be replaced in the previous production tact. A special organic solvent is needed to clean the paint material, which would generate a lot of VOC, which is harmful to the environment and human beings. Hence, minimising the times of colour switching is required, as spray gun cleaning is performed whenever the colour needs to be changed [31]. Moreover, two types of parts can be found in the M2M-AS: standard and key parts. This article only considers the key parts (G) which are affected by the assembly sequence. Smoothing the part demand [32] is obtained in the first function to reduce demand fluctuations in each period. Meanwhile, the assembly load balancing is formulated as a constraint satisfaction problem [33].

To balance the sustainability of economic, environmental and social dimensions, a MOMINLP model is proposed to formulate this problem. Notably, the production also has a major influence on the environment and society. Consequently, not only sustainability from transportation is considered but also the environmental and social effects of production. In the environmental objective, VOC and CO_2 emission costs are counted. The social objective



FIGURE 1. PI-enabled hyperconnected production-distribution system.

considers the social welfare contribution of carbon mitigation and workers' rights from workers' safe and fair production. The sets, parameters and decision variables are defined as the following notation list:

B. MATHEMATICAL DESCRIPTION OF THE SUSTAINABILITY

1) ECONOMIC OBJECTIVE

The economic objective function (1) minimises the sum of fixed setup cost and variable cost for production, inventory cost, loading cost, transportation cost and delivery tardiness cost. The variable production cost includes the colour switching cost for PT and the supply cost for AS.

$$f_{Eoc} = \sum_{t \in T} \sum_{i \in P} \left(\sum_{k \in K} \left(\sum_{a \in O} \tau_{ak} \cdot x_{ai}^{t} - \varphi_{l} \cdot \theta_{k} \right) \cdot C_{q}/q \right)$$

$$+ \sum_{n \in NT} \sum_{i \in P} \sum_{e \in E} \sum_{d \in E} X_{edi}^{nt} \cdot (C_{r} + C_{u})$$

$$+ \sum_{t \in T} \sum_{i \in P} \eta_{i}^{t} \cdot C_{p} + \sum_{t \in T} \sum_{a \in O} \zeta_{a}^{t} \cdot C_{v}$$

$$+ \sum_{t \in T} \sum_{i \in N} \sum_{j \in N \setminus P} \sum_{m \in M} \left(\check{C}_{m} + \widehat{C}_{m} \cdot d_{ijm} \right)$$

$$\cdot \left[\left(\sum_{a \in O} \pi_{aijm}^{t} \right) / S_{m} \right]$$

$$+ \sum_{a \in O} \left(\widehat{C}_{d} \cdot 2 \cdot \sum_{t \in T} \sum_{i \in N} \sum_{j \in N \setminus P} \pi_{aijm}^{t} \right)$$

$$+ \sum_{t \in T} \sum_{i \in N_{1}} I_{i}^{t} \cdot \check{C}_{d}$$

$$+ \sum_{j \in R} \left(\psi_{aj} \cdot \left(\sum_{a \in O} \max \left\{ 0, \sum_{t \in T} \sum_{i \in N} \sum_{m \in M} (t + d_{ij}/\bar{v}_{m}) \right. \right)$$

$$\cdot \pi_{aijm}^{t} - \kappa_{a} \right\} \cdot \rho \right) \right)$$

$$(1)$$

2) ENVIRONMENTAL OBJECTIVE

In the environmental objective, the cost of GHG emissions caused by painting and transportation are considered, as there is no energy consumption reduction during the smoothing part demand for AS. The first and the second terms are GHG consumption for different transportation modes ([34], [35]). Moreover, the third term calculates the process cost of VOC [36].

$$f_{Env} = \sum_{t \in T} \sum_{i \in N} \sum_{j \in N} \left[\sum_{a \in O} \pi^t_{aij1} / \left(\left[\sum_{a \in O} \pi^t_{aij1} / S_1 \right] \cdot S_1 \right) + S_t / \left(\sum_{a \in O} \pi^t_{aij1} \cdot S_f + S_t \right) \right] \cdot d_{ij1} \cdot \delta_1 \cdot R_c + Q_c \cdot \delta_c \cdot \delta_d \cdot \sum_{t \in T} \sum_{i \in N \setminus R} \sum_{j \in N} d_{ij2} \cdot \left[\sum_{a \in O} \pi^t_{aij2} / S_2 \right] \cdot R_c + \sum_{nt \in NT} \sum_{e \in E} \sum_{d \in E} \sum_{i \in P} X^{nt}_{edi} \cdot Q_v \cdot R_v$$
(2)

3) SOCIAL OBJECTIVE

Regarding social objective, the objective function (3), as shown at the bottom of the page, is evaluated from three aspects: the social contribution of carbon mitigation, workers' safety concerns [37] and production fairness [38]. One part of the social contribution of carbon mitigation comes from PI-enabled transportation. The other part comes from the recycling system of VOC in PT, which is generated during the cleaning of nozzles. Moreover, the workers' safety concern is contributed by the cost-saving of colour switching between the traditional PT and M2M-PT [39]. Fair production is calculated based on the Coefficient of Variation for the production period between plants.

C. GENERIC MODEL

The sustainability of PI-HPDS can be presented as min $f_{Sus} = (f_{Eoc}, f_{Env}, -f_{Soc})$. The constraints are as (4)–(18), shown at the bottom of page 7.

The assembly schedule must satisfy the limitations of production capacity, production sequence, and production bottlenecks. Constraints (4) indicate that the order *a* must not be assembled in the plant *i* at the period *t* when the plant *i* does not perform the production at the period *t*. Constraints (4) also restrict the production capacity of AS for every plant in each period. Constraints (5) further stipulate the transfer of semi-finished products between the plants. Constraints (6) define the Coefficient of Variation for fair production. Constraints (7) specify that each order must be produced at most once. The assembly sequence is represented by constraints (8) - (15), where virtual orders 0 and |O| + 1 are set as the head and tail orders for each time period in constraints (9) and (10). Constricts (11) and (12) define the cross-period connections for the assembly sequence. Constraints (13) and

$$f_{Soc} = \vartheta \cdot \omega_l + \omega_s / \sum_{t \in T} \sum_{a \in O} \sum_{i \in P} \mu_{ai}^t + \left(\sum_{\text{nt} \in \text{NT}} \sum_{i \in P} \Upsilon_i^{\text{nt}-1} - \sum_{\text{nt} \in \text{NT}} \sum_{e \in E} \sum_{d \in E} \sum_{i \in P} X_{edi}^{\text{nt}} \right) \cdot Q_u \cdot R_c + \left(\sum_{t \in T} \sum_{i \in N_1} \sum_{j \in N_1} \delta_2 \cdot d_{ij2} \cdot \sum_{a \in O} \pi_{aij1}^t \cdot \varsigma_f - \left[\sum_{a \in O} \pi_{aij1}^t / \left(\left[\sum_{a \in O} \pi_{aij1}^t / S_1 \right] \cdot S_1 \right) + \varsigma_t / \left(\sum_{a \in O} \pi_{aij1}^t \cdot \varsigma_f + \varsigma_t \right) \right] \cdot d_{ij1} \cdot \delta_1 \right) \cdot R_c$$
(3)

TABLE 2. Notation list.

Sets				
W	Set of workshop; $W = \{s, l, h\}$, s, l, h represen	t PT, AS, a	nd CBS respect	ively.
U	Set of Greenhouse gases (GHG); $U = \{c, v\}$, c	c and v repr	resent CO ₂ and '	VOC respectively.
NT	Set of time slot in painting workshop; $nt \in \{1, \dots, n\}$	2,, NT }		
H_t	Set of time slot for every period in assembly v	workshop;	$sl \in \{1, 2,, H_t $	}
Parameters				
1 w	The length of production tact for workshop, V	$W = \{s, l, h\}$	C_u	Set up cost of gun
$arphi_w$	Production capacity of workshop, $W = \{s, l, l\}$	$h\}$	C_p	Fixed product cost per
C_r	Unit cost of organic solvent on gun cleaning c	operation	C_d	Unit (un)loading cost
C_{ν}	Transport cost of semi-finished products betw	een plants		
C_q	Supply cost per hour for line-side supply	ω_{s}	All worker pr	otection costs for year
q	The number of parts on each hour supplied by	v side-by lir	ne supply	
S_l	The number of production stations in AS	S_n	The total num	ber of spray nozzles
K_a	Demand time period of order $a \in O$	R_u	Tax for GHG	, $U = \{c, v\}$
ρ	Unit cost of tardiness	S_i	Storage capac	city of plants and hubs
${\boldsymbol{\varsigma}}_f$	Locomotive weight of finished vehicle	$\boldsymbol{\varsigma}_t$	Locomotive v	veight of train
\dot{c}_{d}	Unit holding cost of plants and hubs	$\delta_{_d}$	Load correcti	on factor
ω_l	The reward of production fair	$\delta_{_c}$	Road gradien	t correction factor
\dot{C}_m	Variable transportation cost of mode <i>m</i>	\dot{C}_m	Fixed transpo	rtation cost of mode m
d_{ijm}	The distance from node <i>j</i> to <i>i</i> in mode <i>m</i>	$\bar{\mathcal{V}}_m$	Average spee	d of mode <i>m</i>
$\delta_{_m}$	Carbon emission factor of each transportation	facility for	transportation	mode <i>m</i>
$q_g: Q_g$	Maximum quantity limit on the continuous p	roduction f	or part g	
θ_{k}	Standard ratio accounted for the production ca	apacity of s	eries k in each p	period
S_m	Capacity of each transportation facility for tra	nsportation	mode <i>m</i>	
$\sigma_{_{ki}}$	$\overline{\sigma}_{ki} = 1$, if series k is stamped in plant i; 0, oth	nerwise		
$ au_{_{ak}}$	$\tau_{ak} = 1$, if order <i>a</i> is an order of series <i>k</i> ; 0, oth	herwise		
ψ_{ar}	$\Psi_{ar} = 1$, if order <i>a</i> is required by retailer <i>r</i> ; 0,	otherwise		
ξ_{ae}	$\xi_{ae} = 1$, id order <i>a</i> is an order of colour <i>e</i> ; 0, o	therwise		
l_{kg}	$l_{kg} = 1$, if part g is installed in series k; 0, othe	erwise		
Q_c	The rate of CO ₂ emissions (gram/kilometre) for	or a truck o	n road with zer	o gradient
Q_{ν}	VOC emissions on each spray gun cleaning of	peration		
Q_u	The rate of CO ₂ emissions for VOC recovery	system		
D_{ijm}^t	$D'_{ijm} = 1$, if mode <i>m</i> is available for the arc <i>i</i> to	o <i>j</i> in period	l t; 0, otherwise	
Decision varial	bles	× •		
x_{ai}^{t}	1, if the assembly of order <i>a</i> is performed at p	lant <i>i</i> in per	riod t; 0, otherw	ise
\hat{x}_{ai}^{slt}	1, if the assembly of order a is performed at p	lant <i>i</i> in tim	ne slot <i>sl</i> during	period t; 0, otherwise
\pmb{lpha}^{t}_{abi}	1, if the assembly of order <i>a</i> is preceding for c	order b at pl	lant <i>i</i> in period <i>i</i>	; 0, otherwise
μ_{ai}^{t}	1, if the painting of order a is performed at pla	ant <i>i</i> in peri	od t; 0, otherwi	se
λ_{ai}^{nt}	1, if the painting of order a is performed at pla	ant <i>i</i> in time	e slot <i>nt</i> ; 0, othe	rwise
$oldsymbol{eta}^{\scriptscriptstyle t}_{\scriptscriptstyle abi}$	1, if the painting of order a is preceding for or	rder <i>b</i> at pla	int <i>i</i> in period <i>t</i> ;	0, otherwise
$\pi^{\scriptscriptstyle t}_{\scriptscriptstyle aiim}$	1. if order <i>a</i> is delivered by mode <i>m</i> from node	e <i>i</i> to <i>i</i> in pe	eriod t: 0. othery	wise

\mathcal{Y}_{ai}^{t}	1, if order <i>a</i> is detained at node <i>i</i> at the end of period <i>t</i> ; 0, otherwise
ζ_a^t	1, if order a is transmitted between plants in period t ; 0, otherwise
\mathbf{X}_{edi}^{nt}	1, if colour e is changed by colour d at plant i in time slot nt ; 0, otherwise
E_e^{nt}	1, if colour e is included by spray package in time slot nt ; 0, otherwise
η_i^t	1, if plant <i>i</i> perform production in period <i>t</i> ; 0, otherwise
I_i^t	Inventory of node <i>i</i> at the end of period <i>t</i>
$\boldsymbol{\chi}_{ai}^{t}$	1, if order a is offline from assembly shop at plant i in period t ; 0, otherwise
f_i^{nt-1}	1, if traditional colour switching is performed on time slot nt -1 in period t at plant i ; 0, otherwise
$Z_{abi}^{t,t+1}$	Indicator variable. 1, if $\beta_{a O +1}^t = \beta_{0a}^{t+1} = 1, \forall a \in O$; 0, otherwise
STA_a	The time that order <i>a</i> starts to be processed in AS
STP _a	The time that order <i>a</i> starts to be processed in PT
ETA _a	The time that order <i>a</i> is completely assembled in AS
ETP _a	The time that order a is completely painted in AS

TABLE 2. (Continued.) Notation list.

(14) calculate the start time and the offline time for each order in AS, where the assembly of each order needs to go through all stations in sequence. Constraints (15) specify that

the specific order produced in the time slot sl during the time period t is determined. Constraints (16) - (18) satisfy the assembly load balancing requirement. Constraints (16)

$$\sum_{a \in O} x_{ai}^t \le \eta_i^t \cdot \varphi_l \quad \forall i \in P, \ \forall t \in T$$
(4)

$$\zeta_a^{t-1} = \left| \tau_{ak} \cdot \left(i \cdot x_{ai}^t - i \cdot \varpi_{ki} \right) \right| \quad \forall a \in O, \ \forall i \in P, \ \forall t \in T$$

$$(5)$$

$$\vartheta = 1 - \frac{\sqrt[2]{\sum_{i \in P} \left(\sum_{t \in T} \eta_i^t - \sum_{t \in T} \sum_{i \in P} \eta_i^t / |P|\right)^2 / |P|}}{\sum_{t \in T} \sum_{i \in P} \eta_i^t / |P|}$$
(6)

$$\sum_{t \in T} \sum_{i \in P} x_{ai}^{t} = \sum_{t \in T} \sum_{i \in P} \chi_{ai}^{t} = 1 \quad \forall a \in O$$

$$\tag{7}$$

$$\sum_{a \in O \cup 0, b \neq a} \alpha_{abi}^t = \sum_{a \in O \cup |O|+1, b \neq a} \alpha_{abi}^t = x_{bi}^t \quad \forall b \in O, \ \forall i \in P, \ \forall t \in T$$
(8)

$$\sum_{a \in O} \alpha_{a0i}^t = 0 \quad \forall i \in P, \ \forall t \in T$$
(9)

$$\sum_{a \in Q} \alpha^t_{|O|+1,a,i} = 0 \quad \forall i \in P, \ \forall t \in T$$

$$\tag{10}$$

$$z_{abi}^{t,t+1} - 1 + \varepsilon \le \varepsilon \cdot \alpha_{a,|O|+1,i}^t \cdot \alpha_{0bi}^{t+1} \quad \forall a, b \in O, a \neq b, \ \forall i \in P, \ \forall t \in T$$

$$(11)$$

$$z_{abi}^{t,t+1} - 1 - \varepsilon \ge (-1 - \varepsilon) \cdot (1 - \alpha_{a,|O|+1,i}^t \cdot \alpha_{0bi}^{t+1}) \quad \forall a, b \in O, a \neq b, \ \forall i \in P, \ \forall t \in T$$

$$STA_b \ge STA_a + \ell_1 - (1 - \alpha_{a,|O|+1,i}^t \cdot \alpha_{0bi}^{t+1}) \cdot M \quad \forall a, b \in O, a \neq b, \ \forall i \in P, \ \forall t \in T$$

$$(12)$$

$$ETA_a = STA_a + S_l \cdot \ell_l \quad \forall a \in O$$
(14)

$$x_{ai}^{t} = \sum_{\mathbf{s} \in H_{t}} \hat{x}_{ai}^{\mathbf{s} \mathbf{l}, t} \quad \forall a \in O, \; \forall i \in P, \; \forall t \in T$$

$$\tag{15}$$

$$\sum_{\text{sl}}^{\text{sl}+Q_g} \tau_{ak} \cdot \iota_{kg} \cdot \hat{x}_{ai}^{\text{sl},t} \le |q_g| \quad \forall a \in O, \ \forall t \in T, \ \forall \text{sl} \in H_t, \ \forall g \in G, \ \forall k \in K$$

$$PK_{ki}^t = \sum \tau_{ak} \cdot \hat{x}_{ai}^{\text{sl},t} \quad \forall a \in O, \ \forall t \in T, \ \forall i \in P, \ \forall k \in K$$

$$(16)$$

$$P\mathbf{K}_{ki}^{t} = \sum_{\mathrm{sl}\in H_{t}} \tau_{ak} \cdot \hat{x}_{ai}^{\mathrm{sl},t} \quad \forall a \in O, \ \forall t \in T, \ \forall i \in P, \ \forall k \in K$$

$$(17)$$

$$\sum_{k \in K} \mathsf{PK}_{ki}^t \le \varphi_l \quad \forall t \in T, i \in P$$
(18)

specify that the maximum number of key parts G selected in a continuously produced Q_g order is q_g . Constraints (16) make sure that the installation of each workstation is completed within the tact. Constraints (17) and (18) restrict the number of orders in each period.

Painting-related constraints are defined as follows. A colour can be painted only if it is in the spray package. Constraints (19) - (24), as shown at the bottom of the next page, illustrate the logical relationship of colour switching in M2M-PT. Constraints (19) specify the replacement of colour e at the time slot nt is performed only when it is in the spray package during nt-1. Constraints (20) specify that if the colour e is replaced by colour d at the time slot nt, then colour d must be included in the spray package at nt+1. Constraints (21) stipulate that if not being switched, the spray package's colours are still used at the next time slot. Constraints (22) ensure the maximum number of colours in the spray package, and constraints (23) restrict that at least one colour should not be switched. Constraints (24) state that order a cannot be painted on time slot nt if colour eof order *a* is not currently in the spray package. Constraints (25), as shown at the bottom of the next page, indicate that an order must be painted and assembled in the same plant. Constraints (26) - (29), as shown at the bottom of the next page, describe the production capacity for PT in each time period t, where the total time period T can also be counted by time slot nt. The production sequence constraints in PT are stated as constraints (30) - (37), as shown at the bottom of the next page. Constraints (38), as shown at the bottom of the next page, specify that order a cannot start to be assembled before the time that painting operations are performed and the transferring time between CBS. Constraints (39), as shown at the bottom of the next page, describe the logic of colour switching in typical PT.

Transportation-related constraints are described above. Constraints (40), as shown at the bottom of page 10, specify that each vehicle must be delivered to a virtual plant or hub after being assembled. Constraints (41), as shown at the bottom of page 10, restrict the number of finished-vehicles that can be delivered. Constraints (42), as shown at the bottom of page 10, forbid orders to revisit any arcs. Constraints (43) - (45), as shown at the bottom of page 10, specify the departure time of orders. Constraints (47) - (51), as shown at the bottom of page 10, define the shipping time and inventory. Notability, the stock of plants is placed in the virtual node P', so it has zero stock.

IV. SOLVING PROCEDURES

As stated above, PI-HPDS is formulated as a MOMINLP model, simultaneously optimising production in the hyperconnected production system and vehicle routing problem in the hyperconnected distribution system. This model would also be NP-hard. The solution representation and the operators of crossover and mutation based on the characteristics of PI-H are proposed for a more effective progeny population generation. NSGAIII is used as the basic framework of the developed algorithm. Moreover, dynamic programming is proposed to get the optimum scheme for the painting subproblem, improving the quality of the solution and the performance of the algorithm. The PI-HPDS proposed in this article is a nonlinear problem, and even if small evolutionary fluctuations in the decision space may lead to significant changes in the target space. Therefore, we simultaneously analyse the population from both decision space and target space and propose adaptive adjustment mechanisms. The adaptive information feedback mechanism and the adaptive reference point-based selection mechanism are adopted into I-NSGAIII, balancing the convergence and the diversity of the algorithm.

A. SOLUTION REPRESENTATION

Solution representation contributes to mapping PI-HPDS's solution space to I-NSGAIII's search space. Since the intermediate production and transportation need to be determined simultaneously, they partition the incidence matrix A = $[\lambda_{ii}], i = 1, \dots, 2 \cdot |T| + 2, j \in N$ into submatrices A_1 , A_2 , and A_3 . Thus we can take $A = (A_1, A_2, A_3)$, in which A_1 , A_2 , and A_3 represent painting sequencing, assembly sequencing, and transportation scheme. Obviously, λ_{1i} and λ_{2j} denotes order at index j of the painting and assembly sequencing, respectively. If the order at index j of A_1 is the same as the order at index k of A_2 , then j and k should be subject to a relation that $j < k - \ell_h + \varphi_s$. The visited route and transportation mode of each order partition the matrix A_3 into T-paired two-rowed submatrices $\overline{A_3^1}, A_3^2, \dots, A_3^T, A_3^n =$ $[\lambda_{ij}], i \in [2n+1, 2n+2], j \in N$ describes the transportation for the order at index j of A_2 in n^{th} period, including the visited route i = 2n + 1 and transportation state i = 2n + 2. A_2 is set as the order index because only the assembly scheme directly affects subsequent transport scheduling. A simple example with the size $T \times N = 4 \times 5$ is as follows:

	4	2	3	1	$\leftarrow A_1$
	4	3	2	1	$\leftarrow A_2$
2 _	3	5	4	3	$\leftarrow A^{l}$
n _{ij} =	-2	2	2	2	× 11 ₃
	8	11	9	7	$\leftarrow A^2$
	-1	1	1	1	3

which describes that order 4 is painted in first and assembly in first. Subsequently, order 4 is delivered through routes 3 and 8.

B. SUBPROBLEM GUIDED DYNAMIC PROGRAMMING

According to the practical production process of PT, spray nozzles can be changed two colours at a time or one colour at a time and the cost involved varies, thus switching scheme affects the incurred costs. Moreover, the digital-driven PT can meet any production requirements of assembly, while the subsequent transport scheduling is only affected by assembly sequencing. We propose dynamic programming to get the optimum change scheme of spray nozzles in PT, which has no negative influence on assembly and transportation. It is beneficial for the quality of the solution and the performance of I-NSGAIII.

First, we reconstruct the painting sequence in each individual to obtain the compact painting sequence. We replace the consecutive identical colours in the sequence as a colour block. Then, let *arr* be the array of colours in the compact production sequence, and P be the set of the index of colour

а

in the compact production sequence, and *S* be the set of spray nozzles, i.e., {1, 2, 3}, and E_i is the set of colours in the spray colour package when the painting process reaches the colour on the index $i. j_1, j_2, k$ respectively represent the vector of available sprays position, idle sprays position and the underworking spray position $\forall j_1, j_2, k \in S. t \in P$ is the time when the sprays nozzle is used again. To simplify the recurrence function, we introduce a function δ^+ (i, j_1, j_2, k, E, t) to calculate the preceding state when the painting process reaches the *i*th colour in compact sequence, with available and unavailable spray position (j_1, j_2, k) , colour package (E), and

$$\sum_{d \in E} X_{edi}^{\text{nt}} \le E_{ei}^{\text{nt}-1} \quad \forall i \in P, \ \forall e, e \neq d, \ \forall \text{nt} \in \text{NT}$$
(19)

$$\sum_{e \in E} X_{edi}^{\text{nt}-1} - 1 + \varepsilon \le (1+\varepsilon) \cdot E_{di}^{\text{nt}} \quad \forall i \in P, \ \forall d \in E, e \neq d, \ \forall \text{nt} \in \text{NT}$$
(20)

$$E_{ei}^{\text{nt}} = 1 - X_{edi}^{\text{nt}} \cdot E_{ei}^{\text{nt}-1} \quad \forall i \in P, \ \forall e, d \in E, e \neq d, \ \forall \text{nt} \in \text{NT}$$
(21)

$$\sum_{e \in E} E_{ei}^{\text{nt}} + \sum_{e \in E} \sum_{d \in E, d \neq e} X_{edi}^{\text{nt}} = S_n \quad \forall i \in P, \ \forall \text{nt} \in \text{NT}$$
(22)

$$\sum_{e \in E} \sum_{d \in E, d \neq e} X_{edi}^{\text{nt}} \le S_n - 1 \quad \forall i \in P, \ \forall nt \in \text{NT}$$
(23)

$$(1 - E_{ei}^{\mathrm{nt}}) + \xi_{ae} - 2 + \varepsilon \le (M + \varepsilon) \cdot (1 - \lambda_{ai}^{\mathrm{nt}}) \quad \forall i \in P, \ \forall e \in E, \ \forall \mathrm{nt} \in \mathrm{NT}$$
(24)

$$i \cdot \sum_{t \in T} \mu_{ai}^{t} = i \cdot \sum_{t \in T} x_{ai}^{t} \quad \forall i \in P, \ \forall a \in O$$

$$\tag{25}$$

$$\sum_{\mathsf{n}\mathsf{t}=1}^{|T|\cdot\varphi_s} \sum_{i\in P} \lambda_{ai}^{\mathsf{n}\mathsf{t}} = 1 \quad \forall a \in O$$
(26)

$$\sum_{a \in O} \lambda_{ai}^{\text{nt}} \le 1 \quad \forall i \in P, \ \forall nt \in \text{NT}$$
(27)

$$\sum_{a \in O} \mu_{ai}^{t} = \sum_{\text{nt}=(t-1) \cdot \varphi_{s}+1}^{t \cdot \varphi_{s}} \sum_{a \in O} \lambda_{ai}^{\text{nt}} \quad \forall i \in P, \ \forall t \in T, \ \forall \text{nt} \in \text{NT}$$
(28)

$$\sum_{a \in Q} \mu_{ai}^{t} \le \varphi_{s} \quad \forall i \in P, \ \forall t \in T$$
⁽²⁹⁾

$$\sum_{e \in O \cup \{0\}} \beta_{abi}^t = \sum_{a \in O \cup \{|O|+1\}} \beta_{bai}^t = \mu_{bi}^t \quad \forall i \in P, \ \forall b \in O, b \neq a, \ \forall t \in T$$
(30)

$$\sum_{a \in Q} \beta_{a0i}^t = 0 \quad \forall i \in P, \ \forall t \in T$$
(31)

$$\sum_{a \in Q} \beta_{|O|+1,a,i}^t = 0 \quad \forall i \in P, \ \forall t \in T$$
(32)

$$Z_{abi}^{t,t+1} - 1 + \varepsilon \le \varepsilon \cdot \beta_{a,|O|+1,i}^t \cdot \beta_{0bi}^{t+1} \quad \forall t \in T, \ \forall a, b \in O, a \neq b, \ \forall i \in P$$

$$(33)$$

$$\begin{aligned} \mathbf{Z}_{abi}^{t,t+1} - 1 - \varepsilon &\geq (-1 - \varepsilon) \cdot (1 - \beta_{a,|O|+1,i}^t \cdot \beta_{0bi}^{t+1}) \quad \forall t \in T, \ \forall a, b \in O, a \neq b, \ \forall i \in P \\ \mathbf{STP}_a &= \mathbf{nt} \cdot \lambda_{ai}^{\mathbf{nt}} \quad \forall i \in P, \ \forall a \in O, \ \forall \mathbf{nt} \in \mathbf{NT} \end{aligned}$$
(34)

$$STP_{a} = \operatorname{In}^{*} \mathcal{K}_{ai} \quad \forall t \in I, \quad \forall a \in O, \quad \forall m \in I \setminus I \quad (35)$$
$$STP_{b} > STP_{a} + \ell_{s} - (1 - \beta_{t,s}^{*}) \cdot M \quad \forall i \in P, \quad \forall a, b \in O, \quad a \neq b, \quad \forall t \in T \quad (36)$$

$$ETP_a = STP_a + \ell_s \quad \forall a \in O$$
(37)

$$STA_a \ge ETP_a + \ell_h \quad \forall a \in O$$
(38)

$$|e \cdot \xi_{ae} \cdot \lambda_{ai}^{\text{nt}-1} - e \cdot \xi_{ae} \cdot \lambda_{ai}^{\text{nt}}| \le M \cdot \Upsilon_{i}^{\text{nt}-1} \quad \forall \text{nt} \in \text{NT}, \ \forall i \in P$$
(39)

the time of next use (t). The function is defined as (52), shown at the bottom of the next page.

The piecewise linear function g(i, E') describes the pollution emission, σ_1 and σ_2 are the pollution emission of changing one or two colours at a time, where $\sigma_2 > \sigma_1 \lor \sigma_2 < 2 \cdot \sigma_1$ as (53), shown at the bottom of the next page.

Then the recurrence function, used to calculate the total emission, can be written as:

$$f(i', j'_1, j'_2, k', E', t') = \min \{f(i, j_1, j_2, k, E, t) + g(i, i', E') | \delta^+(i, j_1, j_2, k, E, t) = (i', j'_1, j'_2, k', E', t') \}$$
(54)

C. ADAPTIVE REFERENCE POINT-BASED SELECTION MECHANISM

I-NSGAIII adopts Das and Dennis' method to create a number *H* of the structured reference points, approximately equal to population size (*N*). The new population P_{t+1} concludes all *k* individuals in $S_t \setminus F_1$ and the remaining |N-k| members from the last front F_1 that are selected according to the selection mechanism. Each individual has one corresponding reference point. The niche count ρ_j of j^{th} reference point may be zero, meaning that no population members are associated with the j^{th} reference point. At the same time, there are some other reference points is associated with more than two members. Especially, early generations of the population are associated with few reference points. Subsequently, when creating the population, individuals associated with the same reference point that are similar to each other may be chosen, resulting in reduced diversity. To solve this problem, this article takes the self-adaptively elimination operator. The details of the adaptive reference point-based selection mechanism are as follows:

D. ADAPTIVE INFORMATION FEEDBACK MECHANISM

In most metaheuristic algorithms, the updating process does not utilise the historical information of individuals in the previous iterations, leading to the loss of some information conducive to population evolution. We perform the information feedback mechanism to select the previous individuals to update the current offspring, regardless of good or bad individuals. The weight of influence is determined based on the fitness function. We further propose the self-adaptively selection strategy for avoiding the premature convergence problem, based on the evolution state. The main idea is to evaluate the distribution of the current decision space using information entropy, which is then adopted to turn the selection strategy.

$$\sum_{m \in M} \sum_{j \in N_1} D^t_{ijm} \cdot \pi^t_{aijm} = \chi^t_{ai} \quad \forall a \in O, \ \forall i \in P, \ \forall t \in T$$

$$\tag{40}$$

$$\sum_{a \in Q} \sum_{i \in N_1} \sum_{m \in M} D_{ijm}^t \cdot \pi_{aijm}^t \le \varphi_l \quad \forall i \in P, \ \forall t \in T$$

$$\tag{41}$$

$$\sum_{t \in T} \sum_{m \in M} D_{ijm}^t \cdot \pi_{aijm}^t \le 1 \quad \forall a \in O, \ \forall i, j \in N, i \neq j$$
(42)

$$\sum_{t \in T} \sum_{i \in N, i \neq j} \sum_{m \in M} D^{t}_{ijm} \cdot \pi^{t}_{aijm} = \sum_{t \in T} \sum_{i \in N \setminus P, i \neq j} \sum_{m \in M} D^{t}_{jim} \cdot \pi^{t}_{ajim} \quad \forall a \in O, \ \forall j \in N \setminus P$$
(43)

$$y_{aj}^{t+\lceil d_{ijm}/\overline{v_m}\rceil} = \sum_{i\in N} \sum_{m\in M} D_{ijm}^t \cdot \pi_{aijm}^t \quad \forall j \in N \setminus P, \ \forall a \in O, \ \forall t \in T$$
(44)

$$y_{aj}^{t} = y_{aj}^{t-1} \cdot (1 - \sum_{i \in N \setminus P} \sum_{m \in M} D_{jim}^{t} \cdot \pi_{ajim}^{t}) \quad \forall j \in N \setminus P, \ \forall a \in O, \ \forall t \in T$$
(45)

$$\sum_{t \in T} y_{ai}^t \ge 1 \quad \forall i \in N_1, \ \forall a \in O$$
(46)

$$\sum_{t \in T} \sum_{i \in N \setminus P, i \neq j} \sum_{m \in M} t \cdot D_{jim}^{t} \cdot \pi_{ajim}^{t} \ge \sum_{i \in N \setminus P, i \neq j} \left(\left(\sum_{t \in T} \sum_{m \in M} t \cdot D_{ijm}^{t} \cdot \pi_{aijm}^{t} + d_{ijm} / \overline{\nu_{m}} \right) + 1 - \left(1 - \sum_{t \in T} \sum_{i \in N \setminus P, i \neq j} \sum_{m \in M} D_{ijm}^{t} \cdot \pi_{aijm}^{t} \right) \cdot M \quad \forall j \in N, \ \forall a \in O$$

$$(47)$$

$$I_i^t = \sum_{a \in O} \chi_{ai}^t - \sum_{a \in O} \sum_{j \in N \setminus P, j \neq i} \sum_{m \in M} D_{ijm}^t \cdot \pi_{aijm}^t \quad \forall i \in P', \ \forall t \in T$$

$$\tag{49}$$

$$I_i^t = \sum_{a \in O} y_{ai}^t \quad \forall i \in N_1, \ \forall t \in T$$
(50)

$$I_i^t \le S_i \quad \forall i \in N_1 \tag{51}$$

We adopt the affinity theory of the immune system to describe the distribution of the population. The population is composed of N individuals having M genes. According to the problem-specific solution representation, we set the set of $S = \{S_1, S_2, S_3\}$ to describe all the elements that occur in the corresponding gene, where S_1 is the set of order, S_2 is the set of the route, and S_3 is the set of transportation state. The population's entropy is calculated as Eq.(55). Eq.(56)calculates the information entropy of the j^{th} gene of all individuals, where $p_{ij} = n_i/N$ is the probability of i^{th} element at j^{th} gene, and n_j is the size of the set associated with the j^{th} gene. Further, the affinity of i^{th} and j^{th} individuals is calculated as Eq.(57), where H(2) is the entropy of i^{th} and *j*th individuals. So the affinity of the population is calculated as Eq.(58), in which $O_P \in [0, 1]$. During the evolution, the population changes from disorder to order and gradually converges. At the early stage of the evolution, the value of Q_P is small and with the process proceeding, the value becomes bigger.

$$H(N) = \frac{1}{M} \sum_{i \in M} H_j(N)$$
(55)

$$H_j(N) = \sum_{i \in S} -p_{ij} \log p_{ij}$$
(56)

$$Q_{ij} = 1/(1 + H(2)) \tag{57}$$

$$Q_P = 1/(1 + H(N))$$
 (58)

The adaptive information feedback mechanism is given in algorithm 2.

E. SUBPROBLEM GUIDED CROSSOVER AND MUTATION OPERATORS

To realise a more effective progeny population generation, we propose a strategy to identify poor variants and improve Inversion_Mutation and Subtour_Exchange_Crossover based on the subproblem's production logic. Considering the online time of painting and assembly, this article adopts the synchronous crossover and mutation to simultaneously update the relative sequence of orders in painting and assembly. Especially for PT, randomly select two positions in the production sequence i, i' to perform a mutation, if the three colour blocks before and after i^{th} position have the same colour of *arr[i*], while do not include colour *arr[i*], the mutation is not allowed. In the crossover process, the intersection segment and the three colour blocks before and after it are updated into a new colour blocks sequence. If there is the same colour in the original or the new local colours blocks sequence, we remove one of them. Next, we compare the length of the two sequences, if the new sequence is longer than the original sequence, no crossover is performed. The strategy is described in Figure 2. Due to the directionality of transportation routes, we carry out Cross_Exchange_Crossover, Or_Opt Mutation, and 2_Opt Mutation while keeping local nodes unchanged. Meanwhile,

$$\begin{split} \delta^{+}(i, j_{1}, j_{2}, k, \mathbf{E}, t) &= \left\{ (i', j'_{1}, j'_{2}, k', \mathbf{E}', t') | i' \in P, k' \in S, j'_{1} \in S - k', j'_{2} \in S - k', j'_{1} \neq j'_{2}, \\ \mathbf{E}'_{j'_{1}} \in E, \mathbf{E}'_{j'_{2}} \in E, \mathbf{E}'_{k'} \in E, t' \in (i', |P|], \\ \text{if } arr[i] &= arr[i + 2], \\ \text{then}(\text{if } i' = i + 1, \text{then}(k' = j_{1}, \mathbf{E}'_{k'} = arr[i], \mathbf{E}'_{j_{1}} = arr[i'], \\ \text{if } t = i + 1, \text{then}(j'_{1} = j_{2}, j'_{2} = k, \mathbf{E}'_{j'_{2}} = \mathbf{E}_{k}), \\ \text{if } t > i + 1, \text{then}(j'_{1} = k, j'_{2} = j_{2}, \mathbf{E}'_{j'_{2}} = \mathbf{E}_{j_{2}}))) \\ \text{if } arr[i] \neq arr[i + 2], \\ \text{then}(\text{if } i' = i + 1, \text{then}(k' = j_{1}, \mathbf{E}'_{k'} = arr[i], \mathbf{E}'_{j_{1}} = arr[i'], \\ \text{if } t = i + 1, \text{then}(j'_{1} = j_{2}, j'_{2} = k, \mathbf{E}'_{j'_{2}} = \mathbf{E}_{k}), \\ \text{if } t = i + 1, \text{then}(j'_{1} = j_{2}, j'_{2} = k, \mathbf{E}'_{j'_{2}} = \mathbf{E}_{k}), \\ \text{if } t = i + 1, \text{then}(j'_{1} = k, j'_{2} = j_{2}, \mathbf{E}'_{j'_{2}} = \mathbf{E}_{k}), \\ \text{if } t > i + 1, \text{then}(j'_{1} = k, j'_{2} = j_{2}, \mathbf{E}'_{j'_{2}} = \mathbf{E}_{j_{2}})) \\ \text{if } i' = i + 2, \text{then}(k' = j_{1}, \mathbf{E}'_{k'} = arr[i], \mathbf{E}'_{j'_{1}} = arr[i'], \mathbf{E}'_{j'_{2}} = arr[i + 1], \\ \text{if } t = i + 1, \text{then}(j'_{1} = k, j'_{2} = j_{2}, \mathbf{E}'_{j_{2}} = \mathbf{E}_{j_{2}})) \\ \text{if } i' = i + 2, \text{then}(k' = j_{1}, \mathbf{E}'_{k'} = arr[i], \mathbf{E}'_{j_{1}} = arr[i'], \mathbf{E}'_{j'_{2}} = arr[i + 1], \\ \text{if } t = i + 1, \text{then}(j'_{1} = k, j'_{2} = j_{2}), \\ \text{if } t = i + 2, \text{then}(k' = j_{1}, \mathbf{E}'_{k'} = arr[i], \mathbf{E}'_{j_{1}} = arr[i'], \mathbf{E}'_{j'_{2}} = arr[i + 1], \\ \text{if } t = i + 2, \text{then}(j'_{1} = j_{2}, j'_{2} = k))) \right\}$$

$$g(i, i', \mathbf{E}') = \begin{cases} 0 & \text{if } i' = i+1, arr[i] \in \mathbf{E}' \\ & \text{if } i' = i+1, arr[i] \notin \mathbf{E}' \\ \sigma_1 & \text{if } i' = i+2, (arr[i] \notin \mathbf{E}' \lor arr[i+1] \in \mathbf{E}') \lor (arr[i] \in \mathbf{E}' \lor arr[i+1] \notin \mathbf{E}') \\ \sigma_2 & \text{if } i' = i+2, arr[i] \notin \mathbf{E}' \lor arr[i+1] \notin \mathbf{E}' \end{cases}$$
(53)



FIGURE 2. Subproblem guided crossover and mutation operators.

Algorithm 1 Adaptive Reference Point-Based Selection Mechanism $(K, \rho_i, \pi, d, Z^r, F_l, P_{l|1}, \varpi)$

Require: K, ρ_j , $\pi(s \in S_t)$, $d(s \in S_t)$, Z^r , F_l , Q_p^t **Ensure:** P_{t+1} 1: k = 12: do 3: $j_{min} = \{j : argmin_{j \in z^e \rho_i}\}$ 4: $j = random(j_{min})$ $I_i = \{S : \pi(s) = j; s \in F_l\}$ 5: 6: if $I_i \neq 0$ then 7: if $\rho j = 0$ then 8: $P_{l+1} = p_{l+1} \bigcup (s : argmin_{s \in I_i} d(s))$ else // calculate similarity of individuals within this reference point 9: if $Q_p^t \leq Q_0$ then 10: 11: $\hat{D}_{i,j} = H_{i,j}$ 12: else $D_i, j = \sqrt{\sum_{m \in M} (|f_m(X_i) - f_m(X_j)|^2)}$ 13: 14: end if 15: if $D_i, j \leq \varpi$ then 16: I_i remove random (X_i, X_i) 17: end if 18: $p_{l+1} = P_{l-1} \bigcup random(I_i)$ 19: end if 20: $\rho_i = \rho_i + 1, F_l = f_l \backslash s$ 21: k = k + 122: else $Z^r = Z^r/J$ 23: end if 24: while $k \in K$

the crossover and mutation operations are performed only between paths but not within the path.

The overview of crossover and mutation operators is given in algorithm 3.

V. EXPERIMENTAL SETUP

A. TEST INSTANCES AND ALGORITHMS

A large dataset is extracted in consultation with industrial managers. Expressly, fifteen instances are selected, including

IEEEAccess

Algorithm 2 The Adapitive Information Feedback Mechanism

Require: \in , Q_0 , N, n, X^t Ensure: X^{l+1} 1: do Calculate Q_p^t for the population X^t of generation t 2: Perform Subproblem-guided crossover and mutation operators with X^t 3: to generate the offspring \hat{X}^{t+1} $\nabla F_{ij}^{l} = f(\hat{x}_{ij}^{l+1}) // x_{ij}^{l}, \hat{x}_{ij}^{l+1}$ are the jth component of the vector x_{j}^{t} and \hat{x}_{i}^{t+1} , where $\forall x_{i}^{t} \epsilon x^{t+1}, \forall \hat{x}_{i}^{t+1} \epsilon \hat{x}^{t+1}$ 4: $\nabla \overline{F}_{i}^{t} = \max_{i} \left\{ \left| f(x_{ii}^{t}) - f(\hat{x}_{ii}^{t+1}) \right| \right\}$ 5: $\nabla \underline{F}_{j}^{l} = \max_{i} \left\{ \left| f(x_{ij}^{t}) - f(\hat{x}_{ij}^{l+1}) \right| \right\}$ $\gamma_{ij}^{l} = \frac{|\nabla F_{ij}^{t}| - \nabla \underline{f}_{j}^{t}}{\nabla \overline{f^{t}}_{j} - \nabla \underline{f}_{j}^{t}}$ 6: 7: 8: if $Q_p^t \leq Q_0$ then 9: else if $\gamma_{ij}^t \leq \in k = rand(l, 2, ..., N)$ 10: 11: 12: else 13: k = i14: end if 15: // Generate the population X^{l+1} for generation t + 116: $X^{T+1} \leftarrow \emptyset$ 17: **for** *i* in *N* **do** $X_i^{L+1} = \alpha 1.\overline{X}_i^{l+1} + \alpha 2.X_k^t, \text{ where } \alpha 1 = \frac{\int_k^t}{\int_i^{t+1} + \int_k^t}, \alpha 2 = \frac{\int_i^t}{\int_i^{t+1} + \int_k^t}, \forall i \in N$ 18: $X^{L+1} \mid X^{t+1}$ 19: 20: end for 21: t = t + 122: while $t \leq T$

the small scales of $24 \times 2 \times 3$, $24 \times 3 \times 5$, $48 \times 2 \times 3$ 3, $48 \times 3 \times 5$, $48 \times 3 \times 8$; the medium scales of $96 \times 3 \times 3$ $5,96 \times 3 \times 8,120 \times 3 \times 8,120 \times 4 \times 15,150 \times 4 \times 15;$ the large scales of $180 \times 4 \times 15$, $180 \times 5 \times 25$, $240 \times 4 \times 15$ 15, $240 \times 5 \times 25$, $360 \times 5 \times 25$. In this study, we selected classic algorithms including NSGAIII [29], MOEA/D [40], SPEA2 [27], and MOCELL [41] as comparison algorithms. To comprehensively evaluate the performance of I-NSGAIII, I-NSGAIII was compared with CPLEX and comparison algorithms. The relative parameters in the comparison algorithms are identical to those in I-NSGAIII. For the stopping condition of these five algorithms, the maximum number of function evaluations (NFE) is set to 50,000. The stopping condition of CPLEX is the maximum time limit of 10800s. All experiments are performed on a PC with an Intel Core-i5 3.30 GHz processor with 4 GB memory, and algorithms are coded in Java with IntelliJ IDEA 2019. The detailed data used in this article can be downloaded at https://pan.baidu.com/s/ 1dV5-KlrOcr-M7b0we6zdzw.

B. PERFORMANCE METRICS

To evaluate the algorithms' performance, we adopt three important multi-objective performance metrics, Inverted Generational Distance (IGD) [42], Hyper Volume (HV) [43] and C-metric [44]. Specifically, C-metric considers the non-dominated sets to assess the dominance of the two algorithms; IGD and HV are used to assess the convergence and distribution of the solutions. We use the Wilcoxon rank-sum test with a 95% confidence level, the nonparametric Kruskal–Wallis test with a 95% confidence level and the Friedman ranking test to determine the statistical significance of the computation results [45], [46].

- C-metric is calculated as $C(A, B) = |\{x_b \in B | \exists x_a \in A, x_a > x_b\}| / |B|$, in which C(A, B) indicates the percentage of the solutions in the set *B* dominated by the solutions in the set *A*. If C(A, B) > C(B, A), then the algorithm to obtain *A* is regarded better.
- IGD is calculated as IGD(PF*, P) = $\frac{\sum_{x \in PF*} \min_{y \in P} ||x-y||}{|PF*|}$, using ||x y|| to calculate the Euclidean distance between solutions x and y. Notability, the true PF of the real instances studied in this article is unknown. Therefore, we aggregated the set of non-dominated solutions obtained by all algorithms during 21 independent runs and set it as the reference PF* for each instance. A smaller IGD value points to the better algorithm performance.
- HV is calculated as $HV(P) = \bigcup_{x \in P} vol(x), vol(x)$ representing the volume of the hypercube that is enclosed by

Require: HP , P_t , p_{pm} , ppc , p_{tm} , p_{tc}
Ensure: U_{t+1}
1: for $i = 0$ to $ PS // PS$ is the size for P_t do
2: Randomly select an individual X_j from HP
3: // Perform the Subproblem guided crossover and mutation in the production
section
4: if <i>random</i> $[0, 1] < p_{pc}$ then
5: Perform <i>Subproblem guided — crossover</i> between production segments
of X_i and X_j
6: end if
7: if <i>random</i> $[0, 1] < p_{pm}$ then
8: Perform Subproblem guided - mutation with production segments of X_j
9: end if
10: // Perform the crossover and mutation in the transportation section
11: if <i>random</i> $[0, 1] < p_{tc}$ then
12: Perform Cross Exchange between transportion segments of X_i , and X_j
13: end if
14: if <i>random</i> $[0, 1] < P_{tm}$ then
15: Randomly select mutation operator R
16: switch R do
17: $case \ or - opt$
18: Perform <i>or</i> — <i>optmutation</i> with transportion segments of X_i
19: $case 2 - opt$
20: Perform 2 — <i>optmutation</i> with transportion segments of X_i
21: end if
22: end for

Algorithm 3 Subproblem Guided Mutation and Crossover Operators

individual x and the reference point, which is set as (1.0, 1.0, 1.0) in this article. The better performance of the algorithm, the greater value of HV.

C. PARAMETERS TUNNING

I-NSGAIII has 5 key parameters, population size (*Ps*), production crossover probability (*Pc*), transportation crossover probability (*Tc*), production mutation probability (*Pm*), and transportation mutation probability (*Tm*). For obtaining the appropriate parameter combination, the standard Design of Experiment (DOE) [47] method was performed on the representative medium instance with the scale of $120 \times 3 \times 8$. Moreover, the mean value of HV is applied as the response value (*RV*). We set these parameters with different levels, as *Ps* =[100, 150, 200, 250], *Pc* =[0.5, 0.65, 0.8, 0.9], *Tc* =[0.5, 0.65, 0.8, 0.9], *Pm* =[0.05, 0.1, 0.15, 0.2], *Tm* =[0.05, 0.2, 0.35, 0.5]. Subsequently, the DOE is performed using the orthogonal array L16 (4⁵) obtained in the IBM SPSS Statistics 26. For each combination, I-NSGAIII is run 21 times independently.

Specifically, the orthogonal array with the corresponding RV is listed in Table 3. We further analyse the extreme difference in RV values between the four levels for each parameter. The larger extreme difference of a parameter implies that the algorithm is more sensitive to the parameter and the parameter is more important. It can be observed that I-NSGAIII is

most sensitive to *Ps* and second to *Pc*. The trends of *RV* with each parameter are shown in Figure 3. Referring to the results of DOE, the key parameters of I-NSGAIII are selected as follows: Pc = 0.9, Pm = 0.15, Tc = 0.9, Tm = 0.20, and Ps = 250. The rest parameter $Q_o = 0.1$ and $\varepsilon = 0.014$ are obtained by the parameter sensitivity analysis in the next subsection.

D. PARAMETER SENSITIVITY ANALYSES

Two parameters in the adaptive information feedback mechanism can be found, including the affinity threshold Q_o and feedback probability ε To verify the most suitable combination of Q_o and ε , we used the Friedman ranking test to investigate the effect of parameter variation on the algorithm.

We set the affinity threshold and random feedback probability as $Q_o = [0.1, 0.12, 0.14, 0.16, 0.18]$ and $\varepsilon = [0.006, 0.008, 0.01, 0.012, 0.014]$, respectively. For each parameter combination, I-NSGAIII is run 21 times on all instances independently. Taking the HV metric as an example, Figure 4 presents the statistical results of the average Friedman ranking. Figure 4(a) shows that when ε is fixed, the algorithm performance deteriorates as Q_o increases. With $Q_o = 0.1$ and $Q_o = 0.12$, the algorithm has a better performance compared to other situations. By fixing $Q_o \in \{0.1, 0.12\}$, Figure 4(b) shows the influence of the random feedback probability ε on the algorithm. INSGAIII performs best with $\varepsilon = 0.014$, regardless of $Q_o = 0.1$ or $Q_o = 0.12$. Thus, INSGAIII in this article adopts $Q_o = 0.1$ and $\varepsilon = 0.014$.

TABLE 3. The orthogonal array.

Europimont number	Parameter level					DV
	Ps	Pc	Pm	Тс	Tm	πv
1	1	1	1	1	1	70.0032
2	1	2	2	2	2	70.794
3	1	3	3	3	3	71.8441
4	1	4	4	4	4	72.069
5	2	1	2	3	4	71.621
6	2	2	1	4	3	72.2932
7	2	3	4	1	2	72.012
8	2	4	3	2	1	72.8986
9	3	1	3	4	2	73.315
10	3	2	4	3	1	72.539
11	3	3	1	2	4	72.606
12	3	4	2	1	3	73.684
13	4	1	4	$\frac{1}{2}$	3	72.3183
14	4	2	3	1	4	72.531
15	4	3	2	4	1	74 34
16	4	4	- 1	3	2	74 803
10			Average RV	7	2	/ 1.005
Parameter level	P_{S}	Pc	Pm	Tc	Tm	
1	71 178	71.814	72 479	72.058	72 445	
2	72 206	72 039	72.610	72.050	72 783	
3	73.036	72.000	72.647	72.151	72 535	
1	73 551	73 416	72.047	73.004	72.333	
T Extreme difference	2 373	1 602	0.412	0.946	0.576	
	2.375	1.002	0.412	0.940	0.370	
74.00- 73.50- 73.00- 72.50- 72.00- 71.50- 71.00- 100.00 150.00 200.00 250.00 Ps	73.50− 73.00− 2 72.50− 72.00− 71.50−	.60 .7	0 .80 .90 Pc	7 7 8 7 7 7 7 7 7 0	2.70- 2.60- 2.50- 2.40- 2.30- 2.20- .05	.10 .15 .20 Pm
73.20- 73.00- 72.80- 72.60- 72.40- 72.20- 72.00- 	70 .80 Te	• A	72.80- 72.70- 72.60- 72.50- 72.40- 72.30- 72.20-	5 .20 .3 <i>Tm</i>	15 .50	

FIGURE 3. Level trends of the parameters (RV).

This is because the algorithm needs pay more attention to exploration in the early stage of evolution. Meanwhile, smaller Q_o and larger ε values make both locating local optimisation and taking the random feedback method to improve the diversity of the offspring population easier. This enables the algorithm to avoid local optimisation.

VI. DISCUSSION

A. EFFECTIVENESS OF DYNAMIC PROGRAMMING AND INFORMATION FEEDBACK MECHANISM

To verify the effectiveness of proposed strategies, in all instances, this comparison was performed between I-NSGAIII, I-NSGAIII without the subproblem-guided dynamic programming (I-NSGAIII-ND) and I-NSGAIII without the adaptive feedback mechanism (I-NSGAIII-NA). Table 3 summarises the mean and standard deviation of the IGD and HV metrics for I-NSGAIII, I-NSGAIII-ND and I-NSGAIII-NA. Moreover, the Wilcoxon's rank-sum test was performed between I-NSGAIII, I-NSGAIII-ND and I-NSGAIII-NA at 95% confidence level. Marks '+', '-' and '~' indicate that the performance of the comparison algorithm is significantly better, worse and similar to I-NSGAIII, respectively. Dark grey and light grey shading mark optimal and suboptimal performance, respectively.

Table 4 shows that, for most instances, I-NSGAIII has the best performance, and I-NSGAIII-NA obtains suboptimal

т		IGD			HV	
In	I- NSGAIII	I- NSGAIII-NA	I- NSGAIII-ND	I- NSGAIII	I- NSGAIII-NA	I- NSGAIII-ND
1	9.71 <i>e</i> -02	1.26e-01	1.44 <i>e</i> -01	7.41 <i>e</i> -01	7.35e-01	6.73 <i>e</i> -01
1	(2.14 <i>e</i> -02)	(2.30 <i>e</i> -02)-	(1.66 <i>e</i> -02)-	(1.54 <i>e</i> -02)	(4.11 <i>e</i> -02)-	(5.01 <i>e</i> -02)-
2	1.23 <i>e</i> -01	1.29e-01	1.39e-01	7.81 <i>e</i> -01	6.82 <i>e</i> -01	6.38e-01
2	(1.35 <i>e</i> -02)	(1.31 <i>e</i> -02)-	(1.60 <i>e</i> -02)-	(4.82 <i>e</i> -02)	(4.81 <i>e</i> -02)-	(3.51 <i>e</i> -02)-
2	9.944 <i>e</i> -02	1.06 <i>e</i> -01	1.66 <i>e</i> -01	7.40 <i>e</i> -01	7.15e-01	6.65 <i>e</i> -01
3	(2.10 <i>e</i> -02)	(1.95 <i>e</i> -02)-	(1.58 <i>e</i> -02)-	(2.01 <i>e</i> -02)	(3.75 <i>e</i> -02)-	(6.05 <i>e</i> -02)-
	1.01 <i>e</i> -01	1.16e-01	1.64 <i>e</i> -01	7.65 <i>e</i> -01	7.43 <i>e</i> -01	6.74 <i>e</i> -01
4	(1.71 <i>e</i> -02)	(1.84 <i>e</i> -02)-	(2.52 <i>e</i> -02)-	(2.72 <i>e</i> -02)	(4.68 <i>e</i> -02)-	(3.94e-02)-
-	1.16e-01	1.25 <i>e</i> -01	1.54 <i>e</i> -01	7.75 <i>e</i> -01	7.36e-01	6.91 <i>e</i> -01
5	(1.67 <i>e</i> -02)	(1.48 <i>e</i> -02)-	(2.41 <i>e</i> -02)-	(3.63 <i>e</i> -02)	(4.86 <i>e</i> -02)-	(2.91 <i>e</i> -02)-
(1.09 <i>e</i> -01	1.08 <i>e</i> -01	1.43 <i>e</i> -01	7.44 <i>e</i> -01	7.17e-01	6.71 <i>e</i> -01
6	(1.62 <i>e</i> -02)	(2.32e-02)+	(1.50e-02)-	(4.03 <i>e</i> -02)	(4.98e-02)-	(4.47 <i>e</i> -02)-
7	1.09 <i>e</i> -01	1.11 <i>e</i> -01	1.39e-01	7.56 <i>e</i> -01	7.23 <i>e</i> -01	6.77 <i>e</i> -01
/	(2.07 <i>e</i> -02)	(1.95 <i>e</i> -02)-	(2.01 <i>e</i> -02)-	(5.15 <i>e</i> -02)	(3.86e-02)-	(5.00 <i>e</i> -02)-
0	1.05 <i>e</i> -01	1.13 <i>e</i> -01	1.61 <i>e</i> -01	7.56e-01	7.45 <i>e</i> -01	6.98e-01
8	(1.72 <i>e</i> -02)	(1.58 <i>e</i> -02)-	(2.56 <i>e</i> -02)-	(2.72 <i>e</i> -02)	(3.96e-02)-	(5.89 <i>e</i> -02)-
0	1.07 <i>e</i> -02	9.96e-01	1.46 <i>e</i> -01	7.15 <i>e</i> -01	7.28 <i>e</i> -01	6.74 <i>e</i> -01
9	(2.07 <i>e</i> -02)	(1.31e-02)+	(2.25 <i>e</i> -02)-	(1.40 <i>e</i> -02)	(4.43e-02)+	(2.80 <i>e</i> -02)-
10	9.87 <i>e</i> -02	1.16 <i>e</i> -01	1.65 <i>e</i> -01	7.70 <i>e</i> -01	7.20 <i>e</i> -01	6.73 <i>e</i> -01
10	(2.24 <i>e</i> -02)	(1.93 <i>e</i> -02)-	(2.39 <i>e</i> -02)-	(3.38 <i>e</i> -02)	(5.08 <i>e</i> -02)-	(3.88 <i>e</i> -02)-
11	1.17 <i>e</i> -01	1.26 <i>e</i> -01	1.62 <i>e</i> -01	7.76 <i>e</i> -01	7.33 <i>e</i> -01	6.82 <i>e</i> -01
11	(1.46e-02)	(1.35 <i>e</i> -02)-	(2.66 <i>e</i> -02)-	(5.24 <i>e</i> -02)	(3.40 <i>e</i> -02)-	(4.81 <i>e</i> -02)-
12	1.19 <i>e</i> -01	1.31 <i>e</i> -01	1.39 <i>e</i> -01	7.75 <i>e</i> -01	6.80 <i>e</i> -01	6.43 <i>e</i> -01
12	(1.79 <i>e</i> -02)	(2.43 <i>e</i> -02)-	(1.51 <i>e</i> -02)-	(4.08 <i>e</i> -02)	(3.68 <i>e</i> -02)-	(4.96 <i>e</i> -02)-
12	9.87 <i>e</i> -02	1.09 <i>e</i> -01	1.50 <i>e</i> -01	7.58 <i>e</i> -01	7.13 <i>e</i> -01	6.57 <i>e</i> -01
15	(2.00e-02)	(2.38e-02)-	(2.26e-02)-	(5.58 <i>e</i> -02)	(3.82 <i>e</i> -02)-	(6.24 <i>e</i> -02)-
14	1.05 <i>e</i> -01	1.23 <i>e</i> -01	1.38 <i>e</i> -01	7.83 <i>e</i> -01	7.05 <i>e</i> -01	6.47 <i>e</i> -01
14	(2.04e-02)	(2.16 <i>e</i> -02)-	(1.76 <i>e</i> -02)-	(5.14 <i>e</i> -02)	(5.22 <i>e</i> -02)-	(3.68 <i>e</i> -02)-
15	9.83 <i>e</i> -02	1.25 <i>e</i> -01	1.42 <i>e</i> -01	7.59 <i>e</i> -01	7.02 <i>e</i> -01	6.57 <i>e</i> -01
15	(1.28e-02)	(1.97e-02)-	(1.95e-02)-	(5.21 <i>e</i> -02)	(3.47 <i>e</i> -02)-	(2.57 <i>e</i> -02)-
+\-\	~	(2\13\0)	(0\15\0)		(1\14\0)	(0\15\0)

TABLE 4. Per	formance comparise	ons of I-NSGAIII,	I-NSGAIII-NA and	I-NSGAIII-ND.
--------------	--------------------	-------------------	------------------	---------------



FIGURE 4. Sensitivity analysis for the affinity threshold and feedback probability.

performance. The Wilcoxon rank-sum test results clearly show that I-NSGAIII significantly outperforms I-NSGAIII-ND and I-NSGAIII-NA, especially as the scale increases. For suboptimal performance, I-NSGAIII-NA obtains more times than I-NSGAIII-ND; hence, I-NSGAIII-NA is more competitive than I-NSGAIII-ND. The above results fully demonstrate that both subproblem-guided dynamic programming and the adaptive feedback mechanism positively impact the proposed algorithm. Subproblem-guided dynamic programming can more significantly improve algorithm performance.

B. COMPARISON WITH EXACT SOLVER

In this subsection, we performed a comparison of the proposed algorithm with CPLEX in small-scale instances. To solve the multi-objective optimisation problem by CPLEX, we adopt the object weighting method and select 30 uniformly distributed weight vectors in the objective space. We then obtain the non-dominated solution set. As the PF of each real-world instance is unknown, we combine non-dominated solutions obtained with each execution of all algorithms to form the approximate PF to calculate performance

In	I- NSGAIII	CPLEX	NSGAIII	MOEA/D	SPEA2	MOCELL
	9.79 <i>e</i> -02	9.80 <i>e</i> -02	1.46 <i>e</i> -01	1.95 <i>e</i> -01	2.51 <i>e</i> -01	2.59e-01
1	(2.17 <i>e</i> -02)	(2.79 <i>e</i> -02)-	(2.01 <i>e</i> -02)-	(3.02 <i>e</i> -02)-	(4.61 <i>e</i> -02)-	(3.41 <i>e</i> -02)-
2	1.16e-01	1.19e-01	1.28 <i>e</i> -01	1.66 <i>e</i> -01	2.23 <i>e</i> -01	2.18e-01
2	(1.34 <i>e</i> -02)	(1.30 <i>e</i> -02)-	(2.83 <i>e</i> -02)-	(3.29 <i>e</i> -02)-	(3.24e-02)-	(4.34 <i>e</i> -02)-
2	9.86 <i>e</i> -02	2.93 <i>e</i> -01	1.56 <i>e</i> -01	1.67 <i>e</i> -01	2.00 <i>e</i> -01	2.26e-01
3	(2.13 <i>e</i> -02)	(3.11 <i>e</i> -02)-	(2.50 <i>e</i> -02)-	(2.91 <i>e</i> -02)-	(4.76 <i>e</i> -02)-	(3.08 <i>e</i> -02)-
	9.60 <i>e</i> -02	4.39 <i>e</i> -01	1.29 <i>e</i> -01	1.86 <i>e</i> -01	2.38e-01	2.05 <i>e</i> -01
4	(1.81 <i>e</i> -02)	(3.52 <i>e</i> -02)-	(2.71 <i>e</i> -02)-	(2.97 <i>e</i> -02)-	(3.48 <i>e</i> -02)-	(4.13 <i>e</i> -02)-
-	1.16e-01	5.06e-01	1.45 <i>e</i> -01	1.69 <i>e</i> -01	1.90e-01	2.24 <i>e</i> -01
5	(1.67 <i>e</i> -02)	(3.11 <i>e</i> -02)-	(2.10 <i>e</i> -02)-	(3.29 <i>e</i> -02)-	(3.53 <i>e</i> -02)-	(4.13 <i>e</i> -02)-
+\-\~		(0\5\0)	(0\5\0)	(0\5\0)	(0\5\0)	(0\5\0)

TABLE 5. Performance comparisons of IGD values.

TABLE 6. The relative gap %P and %N of CPLEX and I- NSGAIII.

T.,		CPL	EX		I- NSGAIII			
In	Average %P	Max %P	Min %P	Average T	Average %N	Max %N	Min %N	Average T
1	1.83	2.47	0.99	8437	1.57	2.05	0.99	38
2	3.32	5.06	2.34	9981	2.80	3.61	2.08	44
3	6.11	6.75	5.58	10800	3.66	4.79	3.13	49
4	9.74	10.20	8.30	10800	6.67	7.25	5.08	53
5	11.29	12.23	10.44	10800	7.03	7.69	6.12	59



FIGURE 5. The obtained approximate Pareto fronts.



metrics. Table 5 summarises the Wilcoxon's rank-sum test results of IGD.

Table 5 indicates that I- NSGAIII outperforms other algorithms for IGD metric. Moreover, as the scale of the problem grows, the quality of CPLEX's solution worsens. The reason is that the number of variables and constraints increases significantly with the size of the problem. Hence, CPLEX would then add more inequality constraints in the model, making it harder to solve.

Table 6 reports three indicators for evaluating the superiority of the solution obtained by I-NSGAIII and CPLEX, including the relative gap %P, %N and the average running time (*T*). %P indicates the gap between the lower and upper bounds obtained by CPLEX, and %N indicates the gap

between the optimal solution obtained by I-NSGAIII and the upper bound in CPLEX.

I-NSGAIII is observed to always obtain a solution that is not worse than CPLEX. For example, in the case of $24 \times 2 \times 3$ instance with socially sustainable optimisation as the highest priority goal, the average and maximum %*P* are 1.83% and 2.47%, respectively, indicating that the solution is excellent [48]. Table 7 visualises the scheduling result. Thus, the feasibility of our established MOMINLP model is verified. In the remaining three cases, I-NSGAIII can achieve a near-optimal solution within the given timeframe. As the scale increases, the advantages of I-NSGAIII become more obvious than CPLEX.

TABLE 7. The optimal solution for the $24 \times 2 \times 3$ instance.

Production subpro	blem									
Times of cleaning g	un		3		Fluctua	ations in pa	rts den	nand		2
The truck for semi-f	inished tra	insport		4	4					
The production sequ	ience:									
Plant 1 2 5	5 14 7	1 11	6 10	4 13	39					
Plant 2 17 2	0 23 16	19 21	8 18	12 22	15 24					
Distribution subpr	oblem									
Parameters about Tr	ansportati	on (Value))							
<i>I</i> ₃ 1	I_4	0 .	I_5	13	I ₆ 1	11 <i>N</i>	I_1	11	M_2	13
The transportation s	cheduling									
Order T1	T2	Т3	T4	Delay	Order	T1	Т2	T3	T4	Delay
1	1-6	6-9		-2	13			1-2/2-5	5-8	0
2 1-2/2-5	5-7			-1	14	1-6	6-8			0
3		1-2/2-5	5-7	-1	15			2-1/1-6	6-9	-1
4		1-6	6-9	-1	16	2-1/1-6	6-9			0
5 1-6	6-8			0	17	2-5	5-8			0
6	1-6	6-9		0	18		2-5	5-8		-1
7 1-1	1-6	6-8		0	19		2-5	5-7		0
8	2-5	5-7		0	20	2-5	5-8			0
9		1-6	6-9	0	21		2-5	5-8		0
10	1-6	6-8		0	22			2-1/1-6	6-9	-1
11	1-2/2-5	5-7		-1	23	2-5	5-7			-1
12		2-5	5-7	-1	24			2-5	5-8	-1

C. COMPARISON WITH EXISTING ALGORITHMS

In this subsection, we performed the comparisons of I-NSGAIII with the other four compared algorithms in all instances. To intuitively show the performance of algorithms, Figure 5 presents approximate Pareto fronts obtained by these five algorithms based on two different scale instances. For both instances, the solutions obtained by I-NSGAIII are better than the other compared algorithms (no matter the quality or quantity of the solutions). Hence, I-NSGAIII can obtain more reasonable and competitive Pareto fronts.

To further evaluate the effectiveness of I-NSGAIII, the Wilcoxon's rank-sum tests of IGD and HV metrics were conducted. Table 8 and Figure 6(a) show that I-NSGAIII achieves the optimal median of IGD in all instances. Table 9 and Figure 6(b) show that I-NSGAIII achieved the best HV values except for the ninth instance. In the ninth instance, the HV performance of I-NSGAIII is similar to MOEA/D but has a smaller variance. Therefore, the convergence and diversity of I-NSGAIII are better.

Table 10 summarises the C-metric performance of comparison algorithms, where I-NSGAIII, NSGAIII, MOEA/D, SPEA2 and MOCELL are denoted by A, B, C, D and E, respectively. We performed the nonparametric

Kruskal–Wallis test with 95% confidence level with Cmetric; Table 10 lists *p* values. Table 10 shows that I-NSGAIII can obtain more non-dominant solutions than other algorithms in most cases. Thus, I-NSGAIII ensures a remarkable convergence while maintaining a justifiable diversity at its approximated Pareto front. Therefore, the effectiveness of I-NSGAIII is verified. These results confirm the ability of I-NSGAIII to solve the problems encountered in the real industry.

D. COMPARISON OF SUPPLY CHAINS' PERFORMANCE

Without loss of generality, the PI-H, PI-S and TS were tested on medium-scale instances. The Pareto solutions are shown to demonstrate their performance of sustainability. Specifically, the operating characteristics of the three SCs are analysed based on key operating cost indicators. Besides, the sustainability of these three SCs is discussed in different structures of market demands and customer scales. The improvement among PI-S, PI-H and TS was estimated by cost-saving $[(C - C_{PI})/C_{PI}] \times 100\%$. Figure 7 shows their approximate Pareto front (green dots, yellow dots, and red dots represent PI-H, PI-S, and TS, respectively).

In	I- NSGAIII	NSGAIII	MOEA/D	SPEA2	MOCELL
1	9.79e-02(2.17e-02)	1.46e-01(2.01e-02)-	1.95e-01(3.02e-02)-	2.51e-01(4.61e-02)-	2.59e-01(3.41e-02)-
2	1.16e-01(1.34e-02)	1.28e-01(2.83e-02)-	1.66e-01(3.29e-02)-	2.23e-01(3.24e-02)-	2.18e-01(4.34e-02)-
3	9.86e-02(2.13e-02)	1.56e-01(2.50e-02)-	1.67e-01(2.91e-02)-	2.00e-01(4.76e-02)-	2.26e-01(3.08e-02)-
4	9.60e-02(1.81e-02)	1.29e-01(2.71e-02)-	1.86e-01(2.97e-02)-	2.38e-01(3.48e-02)-	2.05e-01(4.13e-02)-
5	1.16e-01(1.67e-02)	1.45e-01(2.10e-02)-	1.69e-01(3.29e-02)-	1.90e-01(3.53e-02)-	2.24e-01(4.13e-02)-
6	1.05e-01(1.96e-02)	1.89e-01(2.82e-02)-	1.79e-01(1.95e-02)-	2.43e-01(4.38e-02)-	2.57e-01(3.84e-02)-
7	1.20e-01(2.07e-02)	1.29e-01(2.63e-02)-	1.80e-01(3.53e-02)-	2.14e-01(3.32e-02)-	2.32e-01(3.01e-02)-
8	1.15e-01(2.02e-02)	1.42e-01(2.38e-02)-	1.85e-01(2.87e-02)-	2.03e-01(2.97e-02)-	2.32e-01(3.85e-02)-
9	9.70e-02(2.07e-02)	1.90e-01(3.87e-02)-	1.23e-01(2.16e-02)-	2.30e-01(4.41e-02)-	2.03e-01(4.47e-02)-
10	9.87 <i>e</i> -02(2.24 <i>e</i> -02)	1.28e-01(2.05e-02)-	1.52e-01(2.86e-02)-	2.31e-01(4.39e-02)-	2.54e-01(3.88e-02)-
11	1.17e-01(1.46e-02)	1.31e-01(1.96e-02)-	1.93e-01(3.25e-02)-	2.14e-01(3.29e-02)-	2.02e-01(3.74e-02)-
12	1.19e-01(1.79e-02)	1.47e-01(2.61e-02)-	1.40e-01(2.08e-02)-	2.12e-01(3.80e-02)-	2.54e-01(3.11e-02)-
13	9.87e-02(2.00e-02)	1.65e-01(2.84e-02)-	1.26e-01(2.95e-02)-	2.30e-01(4.47e-02)-	2.13e-01(4.30e-02)-
14	1.11 <i>e</i> -01(2.02 <i>e</i> -02)	1.64 <i>e</i> -01(2.76 <i>e</i> -02)-	1.77e-01(3.73e-02)-	2.43e-01(3.64e-02)-	2.29e-01(4.46e-02)-
15	9.80e-02(1.28e-02)	1.88e-01(2.82e-02)-	1.81e-01(3.66e-02)-	2.02e-01(3.01e-02)-	2.53e-01(4.21e-02)-
	+\-\~	(0\15\0)	(0\15\0)	(0\15\0)	(0\15\0)

TABLE 8. Performance comparisons of IGD values.

TABLE 9. Performance comparisons of HV values.

In	I- NSGAIII	NSGAIII	MOEA/D	SPEA2	MOCELL
1	7.48e-01(1.57e-02)	6.70 <i>e</i> -01(6.61 <i>e</i> -02)-	6.80e-01(6.77e-02)-	4.50e-01(1.04e-01)-	4.51e-01(7.96e-02)-
2	7.74 <i>e</i> -01(4.71 <i>e</i> -02)	6.33e-01(9.19e-02)-	6.28e-01(7.08e-02)-	4.83e-01(8.27e-02)-	4.33e-01(9.47e-02)-
3	7.43 <i>e</i> -01(2.00 <i>e</i> -02)	6.61 <i>e</i> -01(4.93 <i>e</i> -02)-	6.46e-01(7.93e-02)-	4.86e-01(6.74e-02)-	4.40e-01(1.15e-01)-
4	7.67e-01(2.71e-02)	6.72 <i>e</i> -01(9.29 <i>e</i> -02)-	6.24e-01(7.78e-02)-	4.61e-01(7.40e-02)-	4.16e-01(8.81e-02)-
5	7.73 <i>e</i> -01(3.63 <i>e</i> -02)	6.88e-01(5.33e-02)-	6.68e-01(6.85e-02)-	4.39e-01(1.06e-01)-	4.57e-01(1.16e-01)-
6	7.43 <i>e</i> -01(4.07 <i>e</i> -02)	6.68e-01(8.63e-02)-	6.95e-01(7.00e-02)-	4.20e-01(1.02e-01)-	4.34e-01(9.24e-02)-
7	7.53e-01(5.16e-02)	6.73 <i>e</i> -01(4.86 <i>e</i> -02)-	6.55e-01(7.67e-02)-	4.20e-01(7.38e-02)-	4.18e-01(9.07e-02)-
8	7.51 <i>e</i> -01(2.79 <i>e</i> -02)	6.92e-01(9.85e-02)-	6.35e-01(7.60e-02)-	4.31e-01(5.24e-02)-	4.50e-01(1.19e-01)-
9	7.15 <i>e</i> -01(1.40 <i>e</i> -02)	6.67e-01(4.85e-02)-	7.15e-01(6.86e-02)-	4.82e-01(6.59e-02)-	4.07e-01(1.17e-01)-
10	7.70e-01(3.38e-02)	6.67e-01(6.16e-02)-	6.24e-01(7.75e-02)-	4.24e-01(9.17e-02)-	4.44e-01(7.84e-02)-
11	7.78e-01(5.27e-02)	6.78e-01(7.02e-02)-	6.40e-01(7.09e-02)-	4.71e-01(8.22e-02)-	4.55e-01(1.13e-01)-
12	7.79e-01(4.08e-02)	6.35e-01(7.82e-02)-	6.79e-01(7.85e-02)-	4.19e-01(9.41e-02)-	4.50e-01(9.53e-02)-
13	7.55e-01(5.58e-02)	6.55e-01(9.76e-02)-	6.97e-01(7.79e-02)-	4.21e-01(5.73e-02)-	4.48e-01(9.22e-02)-
14	7.76e-01(5.14e-02)	6.41 <i>e</i> -01(8.16 <i>e</i> -02)-	6.58e-01(7.01e-02)-	4.30e-01(7.33e-02)-	4.42e-01(6.70e-02)-
15	7.54 <i>e</i> -01(5.17 <i>e</i> -02)	6.49 <i>e</i> -01(8.10 <i>e</i> -02)-	6.53e-01(7.86e-02)-	4.39e-01(7.93e-02)-	4.56e-01(6.70e-02)-
	+\-\~	(0\15\0)	(0\14\1)	(0\15\0)	(0\15\0)

Figure 7 demonstrates that the sustainability of PI-H in the social dimension is superior to PI-S. To further analyse the PI-H's sustainability advantage on whether PI-H can enhance sustainability performance without the additional cost of PI-S, the ideal point for PI-S is set to compare with PI-H. The economic, environmental and social object of the ideal point is set as the optimal value in Case A, Case B, and Case C, respectively. The optimal values for different optimisation cases are obtained by setting the priority of the corresponding objective function higher than the other two objectives. In this case, the ideal point of PI-S is (976330, 205.2214, 7504.264) marked in black. Moreover, Figure 7 intuitively shows PI-H has multiple solutions, outperforming the ideal point.

To further clarify PI-H's advantages, the cost performance of each operating factor of the different SCs in three optimisation cases was investigated. To show the changing trend clearly, the operating factors are set as the fixed production cost (FPC); the variable production cost (VPC); the transportation cost (STC); the loading cost (SLC); the inventory holding cost (SHC); the production emission cost (SPE); the transportation emission cost (STE); the production social contribution (SPS); the transportation social contribution (STS). The payoff table in Table 11 clearly shows that PI-H achieves the best performance in every optimisation case. Figures 8-10 demonstrate the operating cost composition of each SC in three optimisation cases specifically.

Under the circumstance of A, the total cost of PI-H and PI-S are lower than that in TS 24.48% and 14.22%, respectively. Obviously, STC and the total production cost are also lower than those in TS. In PI-SCs, the M2M-PT greatly reduces the number of gun switching, and the logical constraints from production are relaxed, which provides more options

T.a	A vs B			A vs C			A vs D			A vs E		
	C(A,B)	$C(\mathbf{B},\mathbf{A})$	р	<i>C</i> (A,C)	C(C,A)	р	<i>C</i> (A,D)	C(D,A)	р	C(A,E)	C(E,A)	р
1	0.716	0.164	.021	0.732	0.108	.008	0.858	0.080	.026	0.820	0.084	.030
2	0.728	0.128	.042	0.724	0.156	.029	0.856	0.096	.027	0.808	0.092	.023
3	0.724	0.148	.035	0.680	0.168	.011	0.826	0.090	.024	0.848	0.056	.044
4	0.748	0.128	.044	0.751	0.120	.008	0.830	0.092	.050	0.856	0.108	.026
5	0.756	0.132	.022	0.700	0.172	.032	0.799	0.114	.002	0.836	0.068	.033
6	0.748	0.140	.014	0.748	0.120	.034	0.820	0.102	.006	0.836	0.068	.033
7	0.696	0.152	.032	0.748	0.152	.043	0.840	0.096	.045	0.860	0.096	.010
8	0.680	0.188	.028	0.700	0.196	.044	0.835	0.091	.027	0.840	0.106	.024
9	0.708	0.148	.026	0.752	0.153	.050	0.843	0.104	.042	0.857	0.093	.026
10	0.744	0.166	.050	0.732	0.156	.032	0.822	0.116	.009	0.804	0.128	.033
11	0.748	0.204	.032	0.716	0.184	.041	0.756	0.128	.042	0.848	0.112	.038
12	0.696	0.172	.015	0.732	0.216	.047	0.856	0.080	.016	0.809	0.096	.030
13	0.668	0.176	.011	0.752	0.200	.001	0.804	0.094	.007	0.832	0.104	.020
14	0.748	0.152	.045	0.712	0.136	.011	0.852	0.092	.024	0.841	0.092	.034
15	0.724	0.148	.038	0.712	0.220	.003	0.842	0.108	.020	0.834	0.112	.023

TABLE 10. Performance comparisons of C-metric.

TABLE 11. Payoff table obtained by the three objective optimisation cases.

Cases	Economical impact			Env	ironmental im	pact	Social impact			
	PI-H	PI-S	TS	PI-H	PI-S	TS	PI-H	PI-S	TS	
Case A	859498	976330	1138179	165.3613	242.558	562.1243	12165.49	7223.562	5751.267	
Case B	913003	1056042	1159353	127.6593	205.2214	504.9185	8582.436	7504.015	5751.332	
Case C	930403	1097161	1210217	148.1636	205.9828	507.089	12831.91	7504.264	5751.357	

for transportation. Besides, the total emission costs of both PI-SCs are also lower than TS. However, the GHG emissions from finished-vehicles holding and loading operations in PI-SCs are higher than those in TS. Furthermore, the emission cost from holding and loading operations, and the total cost in PI-H are lower than those in PI-S. PI-H executes the hyperconnected semi-finished transportation system to realise the connection of the material flow between different workshops in different plants and further integrate plants with different locations into a uniform order pool digitally. PI-H reduces the sum production period, the number of switching in PT, the fluctuation in AS, and transportation cost compared with PI-S. However, the VPC of PI-H is higher than that in PI-S, because the unit cost of the inter-plant transport is higher than the unit cost of painting and assembly. Meanwhile, the social influence of PI-H is also significantly better than PI-S and TS. Moreover, PI-H deepens the production's social advantage through the semi-finished transportation, which balances the uneven distribution of order series, colours, etc., and reduces the discrete coefficient of the production system. Compared with PI-S and TS, PI-H performs better in all three aspects of sustainability.

By analysing solutions in environmental-oriented and social-oriented cases (Figures 8 and 9), we can easily show that differences between these three different SCs vary slightly. Whether cases are environment- or social-oriented,

7490

the costs of PI-SCs remain lower than those in TS. Notably, PI-H improves more environmental benefits at a lower cost than PI-S. Conversely, variable production cost for PI-H increases faster than PI-S. PI-H reduces GHG from colour switching by performing a semi-finished transport, bringing greater VPC compared with PI-S. As PI-S can only adjust scheduling according to orders' colour within each plant, it would cause a delay in the offline time of some orders. Subsequently, to ensure timely arrival of orders, we decreased the vehicle utilisation of the overall transportation system. Consequently, STC of PI-S is much higher than that of PI-H. Notably, PI-S still lags far behind PI-H in the social benefits aspect. This is because PI-S cannot adjust the demand fluctuations caused by MTO, resulting in a larger discrete coefficient of production than PI-H. Generally, PI-H can significantly reduce the sustainable pressure of auto enterprises at a lower cost.

E. MTO'S PARAMETER SENSITIVITY ANALYSIS

The market structure and order scale are the two essential features of MTO. This section examines whether these two features influence PI-H's advantages. We performed analyses on nine combinations of three market structures and three instance sizes. Table 12 presents the experimental results.

Table 12 indicates that TS's performance is always worse than PI-SCs. Therefore, the following analysis mainly

 TABLE 12.
 Summary results for sensitivity analysis with nine types of market structure.

	PI-H-A	PI-H-B	PI-H-C	PI-S-A	PI-S-B	PI-S-C	TS-A	TS-B	TS-C
N1G1	859498	127.66	12831.91	976330	205.22	7504.264	1138179	504.92	5751.357
N1G2	1332955	204.98	12629.75	1472692	339.56	7650.09	1571459	571.72	6072.29
N1G3	1767904	262.76	12727.18	1945995	454.76	8182	2051869	734.92	6615.849
N2G1	892032	157.38	12301.05	983026	253.179	8781.176	1048030	389.324	7365.57
N2G2	1287235	233.98	12301.03	1416178	397.18	8515.11	1505072	542.25	7167.88
N2G3	1747877	272.36	12629.58	1855890	521.96	9239.369	1917820	647.3	7876.073
N3G1	866296	128.02	12831.86	988780	166.779	6273.246	1171640	435.937	4108.108
N3G2	1407040	185.98	12906.49	1589238	253.18	7093.393	1708971	522.41	4969.895
N3G3	1849717	215.69	13346.53	2080785	368.36	6505.255	2194687	675.99	4567.996



FIGURE 6. The IGD values and HV values for all comparison algorithms.



FIGURE 7. The approximate Pareto fronts for the three supply chain configurations.

focuses on PI-H and PI-S. It is clear that the economic and social advantages of PI-H over PI-S become more obvious in all customer scales as the complexity of the market structure increases. However, as market structure complexity increases, the environmental benefit of PI-H over PI-S decreases from 37.837%, 41.088%, 47.8196% to 37.794%, 39.633%, 42.219% and 23.239%, 26.5403%, 41,446% respectively. In the same structure, different customer scales cannot affect the social performance of PI-H and PI-S, while economic advantage of PI-H is in a negative correlation with the customer scales. The advantages are decreasing from 11.967%, 9.256%, 12.387% to 9.489%, 9.1049%, 11.464% and 9.152%, 5.820%, 11.105%. This might be because scale economy gradually improves truck utilisation. Conversely, in the same structure, the environmental advantage of PI-H is in positive correlation with customer scales, and the advantage increases from 37.79%, 37.84%, 23,239% to 39.63%, 41.09%, 26.54% and 42.22%, 47.82%, 41.45%, respectively. Increased orders in the same demand classification impose greater scheduling pressure on the assembly constraints in PI-S compared to PI-H. Notably, PI-H always has superiority in all scenarios. Thus, PI-H can address the MTO market with higher sustainable performance.

VII. MANAGERIAL INSIGHTS

In the above discussion, the sustainability performance of PI-H has been fully demonstrated to practitioners. Beginning from the practical management of the OEM, we extract three important management implications:

- For the OEM, adopting PI-H is recommended if they want to realise MTO more sustainably. PI-H breaking the maximum capacity limit of a single plant brings greater single-cycle flexible scheduling capabilities to address random demands, which have an uneven distribution of order series, colours and delivery time. The hyperconnected supply chain framework is also applicable to other OEMs.
- The OEM should adopt PI-H, especially with a more complex market structure. The above discussions have clearly shown that PI-H will obtain greater economic and social benefits from flexible production capacity as market structure complexity increases. On environmental sustainability, if trucks with new energy can be used for transportation, PI-H will perform better.
- Combining the modular concept with hyperconnected semi-finished transportation will have a considerably positive impact on the overall performance of the automobile supply chain. Existing discussions have found that the transportation cost and efficiency of semi-finished transportation have a significant impact on the supply chain's long-term performance.







FIGURE 9. The performance of operational activities in environmental-oriented optimisation case.



FIGURE 10. The performance of operational activities in social-oriented optimisation case.

The PI-enabled, hyperconnected modular system can distribute a wider variety of semi-finished products at a lower cost.

VIII. CONCLUSION

This study investigated the sustainable production-distribution joint scheduling problem in the PI-enabled hyperconnected order-to-delivery system to promote sustainability, which is crucial for the automobile industry. Hence, workshops in different plants were hyperconnected to produce various series of vehicles. Subsequently, the finished-vehicles were delivered to customers through the three-level transportation system by multiple transportation modes. Re-integration was then performed in PI-hub. We formulated the multiple-objective model based on practical features of PI-HPDS with digital-driven MMS, trading off economic, environmental and social dimensions for SSCM.

In this article, we developed I-NSGAIII to solve the sustainable PI-HPDS. Firstly, we tuned the parameters by DOE test and sensitivity analysis. Secondly, we analysed the effectiveness of dynamic programming and information feedback mechanism in the proposed method. Next, we compared I-NSGAIII with CPLEX in the small-scale instances. We then evaluated the performance of I-NSGAIII by the IGD, HV and C-metric in all instances, comparing with the other four comparison algorithms. Moreover, we discussed the sustainable advantages of PI-H under different optimisation preference scenarios. Finally, we performed the sensitivity analysis to determine whether the market structure and customer demand—the two essential features of MTO influence PI-H's advantages. Through computational experiments, we obtained the following results:

- I-NSGAIII ensures a remarkable convergence and reasonable distribution on its approximated Pareto front, providing better results than comparison algorithms.
- In practice, the proposed algorithm can provide satisfactory solutions for decision-makers to help enterprises weaken their energy consumption and thus generate higher profits.
- Results demonstrate that PI-H consistently has superior sustainability performance under MTO. PI-H will provide greater competitiveness in a more complex market structure.
- If automobile companies want to pursue significant sustainable advantages under the MTO, they should establish a PI-enabled hyperconnection production and distribution system.

Although I-NSGAIII outperforms other algorithms in comparison, it can be further improved using learning strategies. In the future, the algorithm performance should train a classification model to select promising solutions. The future direction of this study is to extend the PI-enabled hyperconnected modular application in the closed-loop supply chain and investigate its resiliency in addressing stochastic disruptions brought about by natural disasters.

REFERENCES

- L.-I. Cioca, L. Ivascu, A. Turi, A. Artene, and G. A. Găman, "Sustainable development model for the automotive industry," *Sustainability*, vol. 11, no. 22, p. 6447, Nov. 2019.
- [2] T. Staeblein and K. Aoki, "Planning and scheduling in the automotive industry: A comparison of industrial practice at German and Japanese makers," *Int. J. Prod. Econ.*, vol. 162, pp. 258–272, Apr. 2015.
- [3] E. Oztemel and S. Gursev, "Literature review of industry 4.0 and related technologies," J. Intell. Manuf., vol. 31, no. 1, pp. 127–182, Jan. 2020.
- [4] S. Ji, Y. T. Xue, and G. S. Zhu, "Physical internet-enabled automobile production-distribution joint optimisation with multi-stage workshop," *Kybernetes*, 2022. Accessed: Jan. 19, 2022. [Online]. Available: https:// www.emerald.com/insight/content/doi/10.1108/K-03-2021-0217/full/ html
- [5] H. Tran-Dang, N. Krommenacker, P. Charpentier, and D.-S. Kim, "Toward the Internet of Things for physical internet: Perspectives and challenges," *IEEE Internet Things J.*, vol. 7, no. 6, pp. 4711–4736, Jun. 2020.
- [6] M. Suzanne and M. Benoit, "Introducing the concept of hyperconnected mobile production," in *Proc. 14th IMHRC*, vol. 16, Karlsruhe, Germany, 2016, pp. 1–15.
- [7] Q. Zhang, M. Zhou, C. Li, X. Zheng, K. Wang, and S. Yang, "Case analysis on value creation and sustainable development path of supply chain integrators," in *Proc. 15th Int. Conf. Service Syst. Service Manag.* (*ICSSSM*), Jul. 2018, pp. 1–6.

- [9] P. Chandra and M. L. Fisher, "Coordination of production and distribution planning," *Eur. J. Oper. Res.*, vol. 72, no. 3, pp. 503–517, 1994.
- [10] C. Low, C.-M. Chang, and B.-Y. Gao, "Integration of production scheduling and delivery in two echelon supply chain," *Int. J. Syst. Sci., Oper. Logistics*, vol. 4, no. 2, pp. 122–134, Apr. 2017.
- [11] P. Devapriya, W. Ferrell, and N. Geismar, "Integrated production and distribution scheduling with a perishable product," *Eur. J. Oper. Rses.*, vol. 259, no. 3, pp. 2906–2916, 2017.
- [12] E. Yağmur and S. E. Kesen, "A memetic algorithm for joint production and distribution scheduling with due dates," *Comput. Ind. Eng.*, vol. 142, pp. 106342.1–106342.13, Apr. 2020.
- [13] S. M. Homayouni, D. B. Fontes, and J. F. Gonçalves, "A multistart biased random key genetic algorithm for the flexible job shop scheduling problem with transportation," *Int. Trans. Oper. Res.*, vol. 30, no. 2, pp. 688–716, 2020.
- [14] S. M. J. M. Al-e-hashem, Y. Rekik, and E. M. Hoseinhajlou, "A hybrid L-shaped method to solve a bi-objective stochastic transshipment-enabled inventory routing problem," *Int. J. Prod. Econ.*, vol. 209, pp. 381–398, Mar. 2019.
- [15] F. Aghajani and M. J. M. Al-e-hashem, "A multi-objective mathematical model for production-distribution scheduling problem," J. Ind. Syst. Eng., vol. 13, pp. 121–132, Feb. 2020.
- [16] V. Solina and G. Mirabelli, "Integrated production-distribution scheduling with energy considerations for efficient food supply chains," *Proc. Comput. Sci.*, vol. 180, no. 11, pp. 797–806, 2021.
- [17] B. Montreuil, "Toward a physical internet: Meeting the global logistics sustainability grand challenge," *Logistics Res.*, vol. 3, nos. 2–3, pp. 71–87, 2011.
- [18] T. G. Crainic and B. Montreuil, "Physical internet enabled hyperconnected city logistics," *Transp. Res. Proc.*, vol. 12, pp. 383–398, Jan. 2016.
- [19] I. B. Mohamed, W. Klibi, O. Labarthe, J.-C. Deschamps, and M. Z. Babai, "Modelling and solution approaches for the interconnected city logistics," *Int. J. Prod. Res.*, vol. 55, no. 9, pp. 2664–2684, May 2017.
- [20] T. Chargui, A. Bekrar, M. Reghioui, and D. Trentesaux, "Proposal of a multi-agent model for the sustainable truck scheduling and containers grouping problem in a road-rail physical internet hub," *Int. J. Prod. Res.*, vol. 58, no. 18, pp. 5477–5501, Sep. 2020.
- [21] Z. Hu, R. G. Askin, and G. Hu, "Hub relay network design for daily driver routes," *Int. J. Prod. Res.*, vol. 57, no. 19, pp. 6130–6145, Oct. 2019.
- [22] X.-S. Peng, S.-F. Ji, and T.-T. Ji, "Promoting sustainability of the integrated production-inventory-distribution system through the physical internet," *Int. J. Prod. Res.*, vol. 58, no. 22, pp. 6985–7004, Nov. 2020.
- [23] S.-F. Ji, X.-S. Peng, and R.-J. Lou, "An integrated model for the production-inventory-distribution problem in the physical internet," *Int. J. Prod. Res.*, vol. 57, no. 4, pp. 1000–1017, 2018.
- B. Ramos, C. Alves, and J. V. D. Carvalho, "An arc flow formulation to the multitrip production, inventory, distribution, and routing problem with time Windows," *Int. Trans. Oper. Res.*, vol. 29, no. 1, pp. 526–553, Jan. 2022.
- [25] A. Gharaei and F. Jolai, "An ERNSGA-III algorithm for the production and distribution planning problem in the multiagent supply chain," *Int. Trans. Oper. Res.*, vol. 28, no. 4, pp. 2139–2168, Jul. 2021.
- [26] G. C. Ciro, F. Dugardin, F. Yalaoui, and R. Kelly, "A NSGA-II and NSGA-III comparison for solving an open shop scheduling problem with resource constraints," *IFAC-PapersOnLine*, vol. 49, no. 12, pp. 1272–1277, 2016.
- [27] M. Laumanns, "SPEA2: Improving the strength Pareto evolutionary algorithm," Zürich, Switzerland, Tech. Rep., 2001.
- [28] K. Deb, S. Agrawal, A. Pratap, and T. Meyarivan, "A fast elitist nondominated sorting genetic algorithm for multi-objective optimization: NSGA-II," in *Proc. Int. Conf. Parallel Problem Solving Nature*. Berlin, Germany: Springer, 2000, pp. 849–858.
- [29] K. Deb and H. Jain, "An evolutionary many-objective optimization algorithm using reference-point-based nondominated sorting approach. Part I: Solving problems with box constraints," *IEEE Trans. Evol. Comput.*, vol. 18, no. 4, pp. 577–601, Aug. 2014.
- [30] G. G. Wang and Y. Tan, "Improving metaheuristic algorithms with information feedback models," *IEEE Trans. Cybern.*, vol. 49, no. 2, pp. 542–555, Feb. 2019.

- [31] D. Y. Yu, S. Q. Zhang, X. Y. Shao, S. Q. Liu, and Y. H. Tian, "A gravitylike mechanism for car sequencing problem with multistage sequencing buffer," *Adv. Mater. Res.*, vols. 314–316, pp. 2232–2237, Aug. 2011.
- [32] J. Yu, Y. Yin, and Z. Chen, "Scheduling of an assembly line with a multiobjective genetic algorithm," *Int. J. Adv. Manuf. Technol.*, vol. 28, nos. 5–6, pp. 551–555, Mar. 2006.
- [33] K. Okamura and H. Yamashina, "A heuristic algorithm for the assembly line model-mix sequencing problem to minimize the risk of stopping the conveyor," *Int. J. Prod. Res.*, vol. 17, no. 3, pp. 233–247, May 1979.
- [34] E. Demir, T. Bektaş, and G. Laporte, "A comparative analysis of several vehicle emission models for road freight transportation," *Transp. Res. D, Transp. Environ.*, vol. 16, no. 5, pp. 347–357, Jul. 2011.
- [35] A. Heinold, "Comparing emission estimation models for rail freight transportation," *Transp. Res. D, Transp. Environ.*, vol. 86, Sep. 2020, Art. no. 102468.
- [36] D. B. Guo, H. L. Wang, X. Xu, Y. T. Shen, and Q. Q. Liang, "Reducing VOC and CO₂ emission in automobile painting workshop by production management," *Shanghai Coating*, vol. 1, no. 55, pp. 51–53, 2017.
- [37] X. R. Li, H. H. Li, X. S. Dai, X. H. Song, H. Q. Jiang, Y. J. Gai, and B. Qu, "Dynamic analysis on occupational hazardous level in an automobile coating production line," *Chin. J. Ind. Med.*, vol. 27, no. 5, pp. 385–386, 2014.
- [38] S. B. Ebrahimi, "A stochastic multi-objective location-allocation-routing problem for tire supply chain considering sustainability aspects and quantity discounts," *J. Cleaner Prod.*, vol. 198, pp. 704–720, Oct. 2018.
- [39] Y. Zhang and F. H. Cheng, "Discussion on cost measurement of coating workshop," *Auto. Tech. Mater.*, vol. 2009, no. 10, pp. 33–37, 2009.
- [40] H. Li and Q. Zhang, "MOEA/D: A multiobjective evolutionary algorithm based on decomposition," *IEEE Trans. Evol. Comput.*, vol. 11, no. 6, pp. 712–731, Dec. 2007.
- [41] A. J. Nebro, J. J. Durillo, F. Luna, B. Dorronsoro, and E. Alba, "MOCell: A cellular genetic algorithm for multiobjective optimization," *Int. J. Intell. Syst.*, vol. 24, no. 7, pp. 726–746, 2009.
- [42] E. Zitzler, L. Thiele, M. Laumanns, C. M. Fonseca, and V. G. Da Fonseca, "Performance assessment of multiobjective optimizers: An analysis and review," *IEEE Trans. Evol. Comput.*, vol. 7, no. 2, pp. 117–132, Apr. 2003.
- [43] E. Zitzler and L. Thiele, "Multi-objective evolutionary algorithms: A comparative case study and the strength Pareto approach," *IEEE Trans. Evol. Comput.*, vol. 3, no. 4, pp. 257–271, Nov. 1999.
- [44] B.-B. Li, L. Wang, and B. Liu, "An effective PSO-based hybrid algorithm for multiobjective permutation flow shop scheduling," *IEEE Trans. Syst.*, *Man, Cybern. A, Syst. Humans*, vol. 38, no. 4, pp. 818–831, Jul. 2008.
- [45] V. Kayvanfar, M. Zandieh, and M. Arashpour, "Hybrid bi-objective economic lot scheduling problem with feasible production plan equipped with an efficient adjunct search technique," *Int. J. Syst. Sci., Oper. Logistics*, pp. 1–24, 2022. Accessed: Mar. 24, 2022, doi: 10.1080/23302674.2022.2059721.
- [46] R. Lotfi, Y. Z. Mehrjerdi, M. S. Pishvaee, A. Sadeghieh, and G.-W. Weber, "A robust optimization model for sustainable and resilient closed-loop supply chain network design considering conditional value at risk," *Numer. Algebra, Control Optim.*, vol. 11, no. 2, pp. 221–253, 2021.
- [47] D. C. Montgomery, "Design and analysis of experiments (2nd ed)," J. Amer. Stat. Assoc., vol. 16, no. 2, pp. 394–448, 2000.
- [48] M. Hewitt, G. L. Nemhauser, and M. W. P. Savelsbergh, "Combining exact and heuristic approaches for the capacitated fixed-charge network flow problem," *INFORMS J. Comput.*, vol. 22, no. 2, pp. 314–325, May 2010.



YAOTING XUE received the B.S. degree in business administration from Beijing Information Science and Technology University, China, in 2015, and the M.S. degree in logistics engineering from Northeastern University, China, in 2019, where she is currently pursuing the Ph.D. degree in management science and engineering. Her current research interests include productiondistribution scheduling, carbon emissions, multiobjective optimization, and physical internet.



SHOUFENG JI is currently a Professor with the School of Management Science and Engineering, Northeastern University. His current research interests include sustainability supply chain and physical internet. He is also leading two projects, which was supported by the National Natural Science Foundation of China.



GUOSONG ZHU received the B.S. degree in logistics engineering from Sydney University, China, in 2005. He is currently working as the Senior Manager at BMW, Supply Chain Planning and Innovation Department.



PENGYUN ZHAO received the M.S. degree in logistics engineering from Northeastern University, China, in 2015, where she is currently pursuing the Ph.D. degree in management science and engineering. Her current research interests include supplier selection, physical internet, inventory distribution, disruption risk, resilient, sustainability, and uncertain demand.

...